Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

4-22-2022 11:30 AM

Using conservation genetics to inform reintroduction of the endangered Mottled Duskywing (Erynnis martialis)

Shayla Kroeze, The University of Western Ontario

Supervisor: Keyghobadi, Nusha, *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Master of Science degree in Biology © Shayla Kroeze 2022

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Biodiversity Commons, Ecology and Evolutionary Biology Commons, Entomology Commons, and the Molecular Genetics Commons

Recommended Citation

Kroeze, Shayla, "Using conservation genetics to inform reintroduction of the endangered Mottled Duskywing (Erynnis martialis)" (2022). *Electronic Thesis and Dissertation Repository*. 8513. https://ir.lib.uwo.ca/etd/8513

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

Abstract

Habitat loss and climate change have caused declines in species diversity and abundance globally, including in butterflies which are important components of many ecosystems. Reintroductions are increasingly used to reverse diversity loss but are most effective when informed using genetics. I developed 24 microsatellites and characterized genetic structure and diversity of the endangered Mottled Duskywing (*Erynnis martialis*) in Ontario and neighbouring provinces and states. These were used to inform a planned reintroduction in Ontario. Populations had moderate levels of genetic diversity, however all but the largest populations may be subject to appreciable levels of genetic drift. Populations more than 8 km apart appear to be isolated from each other. My work forms part of a larger effort to achieve the overall recovery of the species in Ontario. Tools I developed may be used to inform future reintroductions of the species, and to monitor status of introduced and extant populations.

Keywords

Species at-risk, reintroduction, Mottled Duskywing, conservation genetics, microsatellites, molecular ecology

Summary for Lay Audience

A growing human population and associated climate changes and habitat losses have led to declines in species across the globe, with some researchers calling this the sixth mass extinction event. Worldwide declines in insect species threaten a "catastrophic collapse of nature's ecosystems". Butterfly species are facing one of the largest of these declines. Butterflies are important as they provide ecological services and are important indicators of environmental health. Re-introducing species to areas where they once occurred is increasingly being used as a strategy for recovering at-risk populations. Many early reintroductions of butterfly species were largely unsuccessful due to a lack of background research and rigorous protocols. The Mottled Duskywing (*Erynnis martialis*) is a medium-sized, brown butterfly listed as Endangered in Ontario, and reintroductions to formerly occupied locations have begun as part of the species Recovery Plan. I developed and used genetic tools to determine metrics of diversity within and among Mottled Duskywing populations in Ontario and nearby locations. Genetic diversity is an indicator of population health; the more variability there is among individuals of a population, the more likely that a population will be able to survive in a changing environment. Populations had moderate amounts of diversity and differed from one another. However, populations within 8 km of one another were not differentiated. These results have informed a reintroduction of the species to Pinery Provincial Park through the selection of the source population based on the levels of genetic diversity in the populations. Moving forward in the recovery goals, the tools I developed can be used to inform future reintroductions through the selection of other potential source populations

and selection of females used in captive rearing programs. Additionally, they can be used to assess the genetic status of both remnant and reintroduced populations.

Co-Authorship Statement

Chapters 2 and 3 of this thesis are planned manuscripts to be submitted to peerreviewed journals. Chapter 2 will be submitted as a primer note.

Both manuscripts resulting from my thesis chapters will be co-authored with my supervisor Nusha Keyghobadi, and collaborative project leads Jessica Linton, Ryan Norris, Gard Otis, and Adrienne Brewster. Chapter 3 will also include Richard Westwood as a co-author.

For all chapters, I contributed to all components of project completion including research conception and design, sample collection, data collection, lab work, genotyping, data analysis and interpretation, and all writing. For all chapters, NK contributed to research conception and design, supervision through all components of the research, and substantial feedback on writing components, as well as research funding. For all chapters, JL, RN, GO, and AB also contributed to study conception and research funding. For chapter 2, sampling was overseen by me, NK, GO, and JL. Full butterfly bodies for microsatellite development were provided by the captive rearing program at the Cambridge Butterfly Conservatory (AB). For chapter 3, sampling was overseen by me, NK, GO, JL, AB, and RW.

Acknowledgments

I would like to express gratitude and thanks to Dr. Nusha Keyghobadi for being such an amazing support through her supervisory role during my degree. I would not have been able to complete this thesis without endless contributions from Dr. Keyghobadi including her extensive knowledge in the field and lab, valuable feedback at all stages of the project, continued support and encouragement, and genuine kindness. Thank you for helping me develop my research skills and become a better scientist.

Thank you to Dr. Gard Otis, Jessica Linton, and Angela Demarse for helping with identification and field work skills, as well as advice and support throughout the project. I would like to thank all who assisted with field work and sample collection including Dr. Gard Otis, Jessica Linton, Brenda Van Ryswyk, Hashveenah Manoharan, Olyvia Foster, Dr. Richard Westwood, Justis Henault, Jacob Gorneau, Emily Trendos, Mitch Gardiner, John Christensen, and Val Deziel, and various field crews. Although I was not able to use them in my thesis, thank you to Christian Schmidt (Agriculture and Agri-Food Canada), Christi Jaeger (AAFC), and Brad Hubley (Royal Ontario Museum) for collection and shipment of historic museum samples. Thank you to the Cambridge Butterfly Conservatory for providing whole butterfly samples especially Adrienne Brewster and Yvonne Young.

I would also like to thank my committee members, Dr. Bryan Neff, Dr. Gard Otis, and Dr. James Voogt for support, valuable advice, and assessment of all thesis-related activities through my degree at Western. I thank my lab mates Mel Lucas, Andrew Chaulk, Kevin Park, and Jessica Jeong for their support and advice regarding project design, research dissemination, and data analysis. Thanks also to my lab volunteer Michela Contursi for general lab maintenance and contribution to *Wolbachia* lab work.

Thanks to all the collaborators on the Ontario Butterfly Species at Risk Recovery team including Western University, Natural Resources Solutions Inc., Conservation Halton, Alderville First Nations, the University of Guelph, the Nature Conservancy of Canada, Kawartha Land Trust, Royal Ontario Museum, Agriculture and Agri-Food Canada, the Ministry of Natural Resources and Forestry, Ontario Parks (Pinery), Toronto Zoo, Environment and Climate Change Canada, Tallgrass Ontario, Northumberland County, Wildlife Preservation Canada, and the Canadian Wildlife Service for countless unique contributions both to my project and to Mottled Duskywing conservation. I am thankful also for the project leads and collaborators Dr. Nusha Keyghobadi, Jessica Linton, Dr. Ryan Norris, Adrienne Brewster, and Dr. Richard Westwood for overseeing the projects related to Mottled Duskywing conservation. Thank you to the organizations that provided access to land supporting Mottled Duskywing Populations including the Nature Conservancy of Canada (Val Dezeil), Meridian Brick (Jack Hewitt), Albany Pine Bush Preserve Commission (Neil Gifford), New York State Department of Environmental Conservation, Michigan Department of Natural Resources (Alicia Ihnken and Blake Gingrich), and Conservation Halton (Brenda Van Ryswyk). Financial support for this project was provided by the National Science and Engineering Research Council of Canada (NSERC), and Western University. Thank you to all who informed locations and flight timing of Mottled Duskywing including Jessica Linton, Dr. Gard Otis, Brenda

Van Ryswyk, Dr. Richard Westwood, Val Deziel, Dr. Jason Dombroskie, Amanda Dillon, and Tim McCabe.

Thank you also to Dr. Andre Lachance, Dr. Natasha Mhatre, Dr. Michelle Belton, and Dr. Paul Mensink for acting as the committee for my proposal assessment, and to Dr. Natasha Mhatre, Dr. Paul Mensink, Dr. Tim Hain, and Dr. Yolanda Morbey for acting as the committee for my thesis defense. I also want to thank Dr. Gard Otis, and Dr. Adam Yates for many letters of recommendations leading to funding and other opportunities and for helping me become a better scientist. Thank you to the Biology Department administrative team for answering questions and helping me coordinate all components of the completion of my Masters.

A very special thank you to my parents, Amy, and Jim Kroeze, for encouraging me to pursue what I am passionate about. I am endlessly grateful for their unconditional love and support throughout my academic journey and my life. I also want to thank Will for helpful advice, love, and support throughout our time in graduate school.

Table of Contents

Abstractii
Summary for Lay Audienceiii
Co-Authorship Statementv
Acknowledgmentsvi
Table of Contentsix
List of Tables xii
List of Figures xiii
List of Appendices xiv
List of Symbols and Abbreviationsxv
List of Software and Software Packagesxvii
Chapter 11
1 General Introduction
1.1 Biodiversity Loss
1.1.1 Butterfly species declines
1.1.2 Genetic diversity as a component of biodiversity
1.2 Population genetic patterns and processes
1.2.1 Population structure and genetic differentiation
1.2.2 Demographic stability and genetic variation
1.2.3 Microsatellites
1.3 Species reintroductions 11
1.3.1 Strategies for source population selection
1.4 Mottled Duskywing16
1.4.1 Reintroduction efforts for Mottled Duskywing

1.5	Thesis	objectives	. 21
1.6	Refere	ences	23
hapte	er 2		. 32
Dev Mo	velopme ttled Du	ent of tetranucleotide and dinucleotide microsatellites for the at-risk uskywing butterfly, <i>Erynnis martialis</i>	32
2.1	Abstra	nct	. 32
2.2	Body.		32
2.3	Refere	ences	40
hapte	er 3		. 41
Pop to i	oulation nform s	genetic analysis of the at-risk Mottled Duskywing (<i>Erynnis martialis</i>) pecies reintroductions	41
3.1	Abstra	ict	. 41
3.2	Introd	uction	42
3.3	Metho	ods	47
	3.3.1	Sample collection	47
	3.3.2	DNA Extraction	52
	3.3.3	Microsatellite genotyping	52
	3.3.4	Data analysis	52
3.4	Result	S	58
	3.4.1	Genetic diversity	58
	3.4.2	Population structure	61
	3.4.3	Demographic stability	68
3.5	Discus	ssion	70
	3.5.1	Genetic diversity	70
	3.5.2	Population structure	72
	3.5.3	Demographic stability	75
	 1.5 1.6 hapte Dev Mo 2.1 2.2 2.3 hapte Popto in 3.1 3.2 3.3 3.4 3.5 	 1.5 Thesis 1.6 Reference hapter 2 Developmed Mottled Dot 2.1 Abstra 2.2 Body. 2.3 Reference hapter 3 Population to inform se 3.1 Abstra 3.2 Introd 3.3 Method 3.3 Method 3.3.1 3.3.2 3.3.3 3.3.4 3.4 Result 3.4.1 3.4.2 3.4.3 3.5 Discuss 3.5.1 3.5.2 3.5.3 	1.5 Thesis objectives 1.6 References hapter 2 Development of tetranucleotide and dinucleotide microsatellites for the at-risk Mottled Duskywing butterfly, Erynnis martialis 2.1 Abstract 2.2 Body 2.3 References hapter 3 Population genetic analysis of the at-risk Mottled Duskywing (Erynnis martialis) to inform species reintroductions 3.1 Abstract 3.2 Introduction 3.3 Methods 3.3.1 Sample collection 3.3.2 DNA Extraction 3.3.3 Microsatellite genotyping 3.4 Data analysis 3.4.1 Genetic diversity 3.4.2 Population structure 3.4.3 Demographic stability 3.5.1 Genetic diversity 3.5.2 Population structure 3.5.3 Demographic stability

	3.5.4 Recommendations for reintroduction	76
	3.6 Conclusion	78
	3.7 References	79
Cł	napter 4	87
4	General Discussion	87
	4.1 Future applications and research	88
	4.2 Conservation implications	90
	4.3 References	93
5	Appendices	95
Cu	urriculum Vitae	104

List of Tables

Table 2-1. Characterization of 24 microsatellite loci in the Mottled Duskywing (Erynnis
martialis)
Table 2-2. Final primer concentrations for PCR amplification of Mottled Duskywing
microsatellites including multiplex PCRs
Table 3-1. Breakdown of Mottled Duskywing samples collected from all known
populations with permission to sample from Ontario, southern Manitoba, eastern New
York, and northeastern Michigan 50
Table 3-2. Mottled Duskywing microsatellite loci showing evidence of deviation from
Hardy Weinberg Equilibrium and null alleles
Table 3-3. Diversity metrics from microsatellite data obtained from Mottled Duskywing
populations
Table 3-4. Pairwise F _{ST} between populations of Mottled Duskywing
Table 3-5. Variance in effective population sizes (Ne) for sampled Mottled Duskywing
populations
Table 3-6. Tests for evidence of bottlenecks in Mottled Duskywing populations 69
Table 3-7. Genetic diversity metrics reported from previous studies of butterfly species
using microsatellites

List of Figures

Figure 1-1. Photo of a Mottled Duskywing (<i>Erynnis martialis</i>) in the wild
Figure 1-2. Mottled Duskywing host plants, New Jersey Tea and Prairie Red Root 18
Figure 1-3. Current and historical distribution of the Mottled Duskywing (Erynnis
martialis) in Ontario
Figure 3-1. Locations of Mottled Duskywing populations sampled for population genetics
analyses
Figure 3-2. An unsampled Mottled Duskywing in the wild that has experienced natural
wing damage
Figure 3-3. The relationship between genetic differentiation (F_{ST}) and log transformed
geographic distance (km) between pairs of populations of the Mottled Duskywing 63
Figure 3-4. Clustering of Mottled Duskywing samples, based on microsatellite data, using
Discriminant Analysis of Principal Components (DAPC)
Figure 3-5. Clustering of Mottled Duskywing samples using STRUCTURE,
STRUCTURE SELECTOR, and CLUMPAK

List of Appendices

Appendix A: Number of observed alleles per microsatellite locus per Mottled Duskywing	5
population in Ontario, New York, Michigan, and Manitoba	5

List of Symbols and Abbreviations

List of Symbols

bp base pairs
°C degrees Celsius
cm ² centimeters squared
km kilometers
mg milligrams
min minutes
mL milliliters
mM millimolar
s seconds
U enzyme unit
μL microliter
μM micromolar
X times
% percent
List of Abbreviations
A _R allelic richness
BSA bovine serum albumin
DAPC Discriminant Analyses of Principal Components
DNA deoxyribonucleic acid
dNTPs deoxyribonucleotide triphosphates
F _{ST} fixation index
GEAGenome Environment Analyses

H _E	expected heterozygosity
H ₀	observed heterozygosity
HWE	Hardy Weinberg Equilibrium
IBD	Isolation by Distance
IBE	Isolation by Environment
IUCN	International Union for the Conservation of Nature
MgCl	magnesium chloride
<i>N</i> e	effective population size
PCR	polymerase chain reaction
SNP	single nucleotide polymorphism

List of Software and Software Packages

List of Software

ArcGIS Pro: https://www.esri.com/en-us/arcgis/products/arcgis-pro/overview

BOTTLENECK:

http://www1.montpellier.inra.fr/CBGP/software/Bottleneck/bottleneck.html

GeneMarker ® HID: https://softgenetics.com/GeneMarkerHID.php

GENEPOP: https://genepop.curtin.edu.au/

Google Earth Pro: https://support.google.com/earth/answer/21955?hl=en

Micro-Checker 2.2: https://micro-checker.software.informer.com/2.2/

NeEstimator: http://www.molecularfisherieslaboratory.com.au/neestimator-software/

PrinSeq: http://prinseq.sourceforge.net/

R version 4.1.3: https://www.r-project.org/

RStudio: https://www.rstudio.com/

STRUCTURE 2.3.4: https://web.stanford.edu/group/pritchardlab/structure_software/release_versions/v2.3.4/ht ml/structure.html

List of R Packages

adegenet: Jombart 2014

diveRsity: Keenan 2017

ecodist: Goslee and Urban 2007

hierfstat: Goudet 2005

pegas: Paradis 2010

Chapter 1

1 General Introduction

1.1 Biodiversity Loss

The Earth is currently experiencing a rapid and widescale loss of biodiversity that poses a multi-faceted threat to humans; biodiversity provides valuable resources such as medicines and new food sources, access to clean water and air, and ecosystem stability in the face of environmental change (Díaz et al. 2006). Prominent scientists have argued that there is no possibility of humans' survival on Earth without the maintenance of a substantial fraction of the existing biodiversity (Hanski 2005). Yet, biodiversity loss is occurring at alarming rates, recent targets set to reduce biodiversity loss have not been met, and these trends are predicted to continue (Waldron et al. 2017). This has led many researchers to conclude that the Earth is entering a sixth mass extinction, with current extinction rates estimated to be at least a hundred times higher than the background rates (Ceballos et al. 2015). This elevation in extinction rate is due, both directly and indirectly, to human activities. It is now increasingly accepted that we have entered a new era called the "Anthropocene", as humans have become the dominant evolutionary force on the planet (Pievani 2014).

Fragmentation, degradation, and loss of suitable habitat due to growing human populations and climate change are cited as two of the major factors that have caused declines in global biodiversity (Oliver and Morecroft 2014). When their habitats are lost, populations of dependent species are also likely to decline (Brown et al. 2018) or become extinct (Jetz et al. 2007). Habitat fragmentation can create small, isolated populations increasing the risk of extinction (Burkey and Reed 2006). Habitat quality, size, and connectivity are all important for the maintenance of species (Isaak et al. 2007).

Additionally, the Intergovernmental Panel on Climate Change (IPCC) predicts significant changes to global temperature and precipitation patterns by the year 2100 (IPCC 2014). Research has shown that climate change has various effects on species' ecology, such as range shifts and reductions, and phenological shifts such as changes to butterfly emergence times (Parmesan 2006). Climate change contributes to extinction when species cannot adapt or migrate rapidly enough to keep pace with the shifting environmental conditions (Feeley et al. 2012). A modelling study of 540 species in 17 clades of terrestrial vertebrates found that temperatures are rising due to anthropogenic climate change faster than most species can adapt (Quintero and Wiens 2013).

Interactions between habitat loss and climate change may exacerbate the impacts of either of these forces in isolation (Mantya-Pringle et al. 2012). For example, populations that have become smaller due to habitat fragmentation may be more susceptible to environmental changes caused by climate change (i.e., the small population paradigm; Brook et al. 2008). Furthermore, habitat fragmentation may limit the ability of species to migrate and shift their range boundaries in response to climate change (Dyer 1995). Most studies indicate that habitat loss is a larger threat to global biodiversity (Mantya-Pringle et al. 2012). However, the impact of climate change is predicted to worsen (Lemoine et al. 2007; Pimm 2008). Additionally, modelling predicts that future climate change will impact species that are largely unaffected by human activities to date (Pimm 2008).

1.1.1 Butterfly species declines

Insects are the most diverse group of organisms on the planet; they represent more than half of the known terrestrial species in the animal kingdom (Stork 2018). Moreover, it is estimated that there are millions of species of insects that have yet to be described (Wilson 1987). Humans rely on insects for essential services such as decomposition, pollination, pest control, and more (Losey and Vaughan 2006). Worldwide declines in insect species threaten a "catastrophic collapse of nature's ecosystems" (Sánchez-Bayo and Wyckhuys 2019). Despite the vital role that invertebrates play in the function of ecosystems, these declines have received significantly less attention than those of mammals and other charismatic megafauna (Prather et al. 2013). Additionally, past research has shown that of thousands of insect extinctions predicted to have occurred, only 70 of them were documented (Dunn 2005). Many insect extinctions may go unnoticed due to a lack of monitoring, scientific description, and identification of invertebrate species (Eisenhauer et al. 2019).

Butterflies are experiencing some of the largest known declines among the insects, with more than half of Lepidoptera species that have been surveyed declining worldwide (Sánchez-Bayo and Wyckhuys 2019). A study by Dirzo et al. (2014) reported strong evidence of a 35% decline in abundance of Lepidoptera species over 40 years. The leading causes of butterfly declines are habitat loss/degradation, chemical pollution, and climate change (Warren et al. 2021). Previous studies have found that the ranges of butterfly species have been retracting as climate has deteriorated, in some cases leading to population-level extinctions (Thomas et al. 2006). Other threats include exotic plant and animal invasion (Wagner and Van Driesche 2010), natural system modifications

(Geyle et al. 2021), and genetic factors associated with small populations (Schmitt and Hewitt 2004). Conservation strategies that have been implemented aimed at protecting butterflies include restoration of habitat, captive rearing, and reintroductions (Schultz et al. 2008).

Butterflies are important components of ecosystems and often thought of as bioindicators due to their visibility and sensitivity to changes in the environment (Pe'er and Settele 2008). Many consider butterflies the 'poster child' of the invertebrate world (Schultz et al. 2008). They are commonly used as a model species in ecological studies and have been used as indicators of habitat quality (Uehara-Prado and Freitas 2009), species richness (Fleishman et al. 2005), and climate change impact (Vickery 2008). Additionally, they have provided foundational information on metapopulation dynamics which have had important implications for conservation (Hanski et al. 1995; Ehrlich 1992). Butterflies have been found useful as an 'umbrella taxon', where protection of butterflies may ensure the protection of other threatened organisms in the same ecosystems (Betrus et al. 2005). In this thesis, I undertake a population genetic study of the only endangered butterfly in the province of Ontario, the Mottled Duskywing, *Erynnis martialis* (Scudder 1870), to information conservation and restoration efforts for this species.

1.1.2 Genetic diversity as a component of biodiversity

The International Union for the Conservation of Nature (IUCN) recognizes three levels of biodiversity: ecosystems, species, and genetics (McNeely et al. 1990). Often, conservation work focuses on the ecosystem and species levels because they are more visible. Additionally, ecosystem services such as crop pollination make the conservation of species and ecosystems attractive to funding agencies (Spangenberg and Settele 2010). However, genetic diversity, defined as heritable variation among individuals and populations within a species (Rao and Hodgkin 2002), is also of critical importance. Genetic diversity allows populations and species to adapt to environmental change (Lande and Shannon 1996), such as emerging diseases (O'Brien and Evermann 1988), climate change (Ehlers et al. 2008), and other disturbances (Hughes and Stachowicz 2004). Furthermore, a lack of genetic diversity can lead to dire problems in populations such as inbreeding depression (Charlesworth and Charlesworth 1999), which I discuss in detail later. Importantly, inbreeding depression can lead to increased risk of extinction, making it extremely important for the conservation of species (Saccheri et al. 1996; Charlesworth and Charlesworth 1987).

1.2 Population genetic patterns and processes

Population genetics is defined as the study of genetic variation within a locus or loci, among individuals of a population or populations, and provides the conceptual and analytical framework for understanding the causes and consequences of loss of genetic diversity (Hedrick 2011). Genetic variation is influenced by four main genetic processes: natural selection, mutation, gene flow, and genetic drift (Star and Spencer 2013; Wright 1931). Natural selection is a non-random difference among individuals in fitness, due to advantages (or disadvantages) of heritable and variable traits in a particular environment and leads ultimately to an increase in the frequency of beneficial alleles (or decrease in detrimental alleles) from one generation to the next (Gregory 2009). When similar

selection pressures are acting on different populations, those populations will tend to become more similar in allele frequencies (Hendry et al. 2002). Similarly, populations experiencing different selection pressures are likely to diverge in allele frequencies. A mutation is a change in DNA sequence that can result from DNA replication errors, exposure to radiation or mutagens, or infection by viruses (Brown 2002). Selection acts on the variation that is introduced by mutations, such that in any given context mutations can be advantageous and increase fitness, deleterious and decrease fitness, or neutral (Loewe and Hill 2010). Gene flow is the movement of genes from one population to another (McDermott and McDonald 1993). Gene flow between two populations can introduce new genetic variation into a population and causes those populations to become more similar in allele frequencies (Slatkin 1985). Genetic drift is a process by which the frequencies of alleles in a population change by chance over time, due to the random sampling of alleles from a parental population to produce the offspring generation (Charlesworth 2009). Genetic drift is more pronounced the smaller a population is, and it both reduces the amount of genetic variation within populations and, on average, increases divergence among populations (Andrews et al. 2017).

1.2.1 Population structure and genetic differentiation

Population genetic structure is the amount and distribution of genetic variation within and among different populations (McDermott and McDonald 1993). While genetic diversity within populations is critical for both evolution and conservation, genetic variation or differentiation among different populations is also key because it determines whether populations are linked or may be on separate evolutionary trajectories (Duminil et al. 2007). Genetic structure and differentiation among spatially separated populations can be shaped by geographic (e.g., isolation by distance; IBD), environmental (e.g., isolation by environment; IBE), and historic factors (e.g., historical distribution of populations and habitats) (Xu et al. 2017). Both ecological and geographical factors can reduce gene flow between populations leading to divergence (Wang et al. 2013).

Isolation by distance is a positive correlation between genetic and geographic distances of populations, and results when dispersal and gene flow become increasingly limited as the spatial separation of populations increases. Dispersal is the movement of individuals from one population to another (Broquet and Petit 2009) and leads to gene flow if an individual or its offspring breeds with a member of the population to which it has moved (Deere et al. 2017). When populations are isolated by habitat fragmentation and degradation, and individuals are unable to disperse between populations, the populations experience decreased gene flow (Miles et al. 2019). Decreased gene flow, acting in combination with genetic drift, is expected to reduce genetic diversity within the populations and increase genetic differentiation among them (Munshi-South et al. 2016). When gene flow among populations becomes so reduced that it can no longer counter the effects of genetic drift, patterns of IBD also break down (Hutchison and Templeton 1999).

Isolation by environment is a correlation between genetic distance and environmental dissimilarity (Sexton et al. 2014). Isolation by environment can occur when populations occupy different environments that generate divergent selection pressures, which reduces gene flow between those populations (Wang and Bradburd 2014). Isolation by distance and environment are particularly important to understand for at-risk species as they can inform conservation strategies, specifically those directed at managing genetic diversity (Xu et al. 2017).

1.2.2 Demographic stability and genetic variation

Broadly defined, demography is the study of the characteristics of populations, and of factors that influence population growth or decline (Tarsi and Tuff 2012). Demographic parameters relevant to population genetics include population size, survival rates, age structure, sex ratio, density, population growth rates, historical population size changes, and exchange of individuals among populations (Tarsi and Tuff 2012; Shen et al. 2019; Lowe et al. 2017; Hughes et al. 2017). These demographic parameters can affect variation in individual fitness, as well as drift and gene flow, in turn determining patterns of genetic variation. Conversely, observed patterns of genetic variation within and among populations can be used to make indirect inferences about these demographic parameters, and the processes they represent, thereby providing insight into the risk of extinction faced by a given population (Tarsi and Tuff 2012).

1.2.2.1 Effective population size

The most fundamental demographic parameter is the number of individuals in a population, also known as the census population size (N; Lebreton et al. 1992). The effective population size (N_e) is a related and central concept in population genetics and is defined as the number of individuals in an idealized population (i.e., each individual makes an equal reproductive contribution to the next generation) that experiences the same level of genetic drift (i.e., the same effect of random sampling of alleles) as that in the actual population (Kimura and Crow 1963; Plutysnki 2007). Effective population

sizes are often much smaller than census population sizes (Husemann et al. 2016) due to differential reproductive success among individuals, unequal sex ratio, and fluctuating population size (Palstra and Ruzzante 2008). Effective population size reflects how strongly drift acts in a population, and therefore how vulnerable the population is to loss of genetic diversity over time.

1.2.2.2 Demographic bottlenecks

A demographic or population bottleneck is an event that drastically, but temporarily, reduces the census size of a population. Demographic bottlenecks can cause a decrease in the gene pool due to loss of alleles (Wright 1951). This can occur even if a population experiences only a brief period of reduced size (Maruyama and Fuerst 1985), although the loss of allelic diversity is greater the longer and more drastically the census population size is decreased. Detecting recent population bottlenecks is relevant for atrisk species as bottlenecks not only increase the risk of population extinction due to decreased genetic diversity (Luikart et al. 1998) but may also reflect increased vulnerability to demographic stochasticity.

1.2.3 Microsatellites

The genetic variation within and among populations can be quantified and described with the use of molecular or genetic markers. One of such markers, microsatellites, are tandem repeats of short (1-6 base pairs) DNA sequence motifs that are located randomly throughout the genome and thought not to perform any function (i.e., are non-coding DNA) (Vieira et al. 2016, Ellegren 2004). Different alleles at a microsatellite locus have a different number of repeats of the motif. Microsatellites have high mutation rates that often arise through replication slippage (Bhargava and Fuentes 2010) and are therefore highly variable. Because they are non-coding, microsatellites are also likely to be selectively neutral. Additionally, they are bi-parentally inherited and co-dominant meaning that both maternal and paternal DNA are represented, and all homozygote genotypes can be distinguished from heterozygote genotypes (Sah et al. 2021). These characteristics make microsatellites useful for determining genetic diversity and population structure, as well as assessing identity, population of origin, paternity, and kinship of individuals (Chistiakov et al. 2006).

Microsatellites have been used extensively to inform conservation programs (Hedrick 2001), including in Lepidoptera such as the endangered Karner Blue Butterfly, for which microsatellites were used to analyze heterozygote deficiencies, geographic variation, and population structure (Anthony et al. 2001). They are particularly useful for studies of endangered species, as they can be analyzed from very small amounts of tissue or other biological material that can be collected with minimal sampling effort and disturbance to the species (Hansen et al. 2001).

Only 12 whole butterfly samples from a single population that had never been described genetically were available for marker development for this project (see Chapter 2). Therefore, microsatellites were a practical option considering their high variability. While I also considered using single nucleotide polymorphisms (SNPs), the potential for low diversity among the samples available for marker development meant a risk of not locating enough SNP loci to conduct robust analyses, as well as potential biases when applying those loci to other populations.

1.3 Species reintroductions

Unprecedented loss of species has led to an imperative to conserve the biodiversity that remains. Reintroduction is one approach to conserving biodiversity and is defined by the IUCN as "the intentional movement of an organism into a part of its native range from which it has disappeared or become extirpated in historic times" (IUCN 1987). Human-mediated species reintroduction or translocation is increasingly used as a strategy for population recovery, including for at-risk lepidopteran species (Daniels et al. 2018). Although much less effort has gone into invertebrate reintroductions compared to mammals (Jule et al. 2008, Jourdan et al. 2019), examples of the former are becoming more common (Schultz et al. 2008; Andersen et al. 2014; Dumeier et al. 2020; Kelemen and Rehan 2021). The goal of reintroduction is to establish a new population that is selfsustaining, has a low probability of extinction (Fraser 2008), and requires minimal longterm management (IUCN 1998). However, there is still no general agreement on the criteria for a successful reintroduction (Gusset 2009). Some suggested metrics of success are survival of the release generation, breeding by the release generation, and persistence of the re-established population (Seddon 1999).

Early reintroductions were often unsuccessful due to a lack of monitoring and documentation, and a lack of consideration or management of key factors such as habitat restoration and predators (Short et al. 1992). The IUCN created a set of guidelines for reintroductions to reduce the number of failed reintroductions (IUCN/SSC 2013). The guidelines recommend considering feasibility assessments and collection of sufficient background knowledge on the species and emphasize comprehensive monitoring and

documentation at all stages. Genetic monitoring has also become a useful tool in determining efficacy of reintroduction programs; using population genetics, changes in demographic parameters such as genetic structure and differentiation can be investigated and quantified (De Barba et al. 2010). The IUCN guidelines also point to an important role for genetic information and considerations that include adequate genetic diversity in both founders and the population(s) from which they are sourced, balancing the risks of inbreeding versus outbreeding among founders, and genetic provenance or match to the new location or environment (IUCN/SSC 2013).

1.3.1 Strategies for source population selection

1.3.1.1 Genetic considerations for reintroduction: inbreeding depression

If founders for a new population are chosen from a source population that is small, or if too few individuals are introduced, the new population will have a small effective size and low genetic diversity (Tracy et al. 2011). This in turn will result in inbreeding, defined as mating of individuals that are genetically similar to each other (Charlesworth and Charlesworth 1999). Inbreeding increases homozygosity, which can lead to an increased likelihood of deleterious recessive alleles persisting in homozygous form (Lynch 1991) and a decrease in variants that are maintained by heterozygote advantage (Hedrick and Garcia-Dorado 2016). These factors, in turn, can reduce fitness parameters such as survival and longevity (Hansson et al. 2001). This reduction in fitness is termed inbreeding depression and has been shown to increase likelihood of population extinction (Saccheri et al. 1996, Charlesworth and Charlesworth 1987).

Inbreeding depression is a concern in the management of endangered species as these species frequently have small, fragmented populations (Hedrick & Kalinowski 2000). Isolated populations that are experiencing little gene flow may have decreased fitness from the negative impacts of increased levels of genetic drift (Miles et al. 2019). While it is also possible for small, inbred populations to purge deleterious recessive alleles, resulting in a population that is well adapted to their current environment, this may result in the population being less capable of adapting to changes in the environment (van der Valk et al. 2021). Additionally, genetic purging occurs over long evolutionary time frames and is largely observed in species with historically small population size and low genetic diversity. Inbreeding depression is of particular concern in reintroduced populations because of the founder effect, which is "the establishment of a new population by a few original founders... which carry only a fraction of the genotypes of the parental population" (Mayr 1963). This effect promotes inbreeding and can enhance the rate of divergence from other populations in the species, which is more pronounced in short-lived species such as insects (Templeton 1980).

Research has shown that inbreeding can have different effects depending on the organism. For example, some invertebrates can thrive at levels of inbreeding that would result in inbreeding depression in mammals (Haikola et al. 2001). However, inbreeding has contributed to the extinction of wild butterfly populations (Frankham and Ralls 1998). Inbreeding has also been shown to have negative impacts on fitness of a butterfly species bred in captivity, with effects such as large decreases in egg hatching (Saccheri et al. 1996) and ability to produce offspring (Franke & Fischer 2013). Similar effects were

seen in a large, natural butterfly metapopulation with negative impacts on larval survival, adult longevity, and egg-hatching rate (Saccheri et al. 1998).

Selecting multiple source populations for reintroduction can increase gene pool variability and avoid inbreeding, but it is important to consider whether individual sources are sufficiently similar to be able to mate and avoid outbreeding depression (McClelland and Naish 2007), which is a reduction in fitness due to mating of highly dissimilar genotypes (Lynch 1991). Inbreeding depression and outbreeding depression can have equally damaging effects on the health of populations (Edmands 2007). However, outbreeding has been found to be less likely to occur than has been historically predicted (Frankham et al. 2017; Weeks et al. 2017; Kronenberger et al. 2018). A review of empirical data indicated that most matings among individuals from distant populations failed to develop outbreeding depression, even if those populations had been isolated for thousands of generations (Frankham et al. 2011). The probability of outbreeding depression is elevated when populations exhibit at least one of the following: mating occurs between two distinct species, the populations have fixed chromosomal differences, have not exchanged genes in the last 500 years, or inhabit very different environments. While the potential for inbreeding depression might be higher for most species, the effects of both inbreeding and outbreeding can be detrimental and therefore it is important to consider both.

1.3.1.2 Genetic considerations: Match to the local environment

Differences in habitat conditions can exert selection pressures that result in genotypes specifically adapted to the local landscape and climate, a condition known as local adaptation (Savolainen et al. 2007). Local adaptation of populations is an important consideration for population reintroduction as it may limit success of reintroduced individuals if environmental conditions in the reintroduction site differ from those experienced by the source population (Lesica and Allendorf 1999). The adaptive potential and pre-existing adaptation strategies provide two approaches to using genetic information to increase the chance that an introduced population will be suited to conditions of the proposed reintroduction site (Houde et al. 2015). The adaptive potential strategy builds upon the theory that the rate of adaptation in a population is proportional to the genetic diversity (Houde et al. 2015); this strategy aims to maximize genetic diversity of the reintroduced population, by selecting diverse founders, to facilitate adaptation to the new environment. In contrast, the pre-existing adaptation strategy aims to maximize the initial local adaptation of the reintroduced population by selecting founders from a source population that experiences environmental conditions matching the reintroduction site.

There is some debate regarding which of these two strategies is more effective. A comprehensive review recommends that the pre-existing adaptation strategy be considered first, as even if a founding group has high genetic diversity, individuals in that group will likely have low fitness if they are not pre-adapted to the new environment (Houde et al. 2015). However, this strategy is specifically recommended, and most effective, when key environment features affecting individual fitness have been

identified. In cases where the key environment and habitat features affecting individual fitness have not yet been clearly identified, the adaptive potential strategy may be the most practical option. Selecting founders to maximize adaptive potential will also be beneficial if environmental conditions at the reintroduction site change over time (e.g., due to climate change).

1.4 Mottled Duskywing

The Mottled Duskywing Erynnis martialis (Scudder, 1870) (Lepidoptera: *Hesperiidae* Figure 1-1) is a medium-sized, brown coloured butterfly belonging to the skipper subfamily Pyrginae, and is endemic to North America (COSEWIC 2015). Adults fly in the summer, in one or two separate broods depending on geographic range. Following fertilization, female butterflies lay their eggs on one of two larval food plant species, New Jersey Tea (Ceanothus americanus) and Prairie Red Root (Ceanothus herbaceus) (COSEWIC 2015; Figure 1-2). The larvae feed on the leaves of these plants and following maturation, either pupate and emerge as butterflies where there is a second brood or overwinter as a mature larva in a winter leaf nest before pupating and emerging the next spring. Mottled Duskywings are believed to disperse less than other Duskywing species based on their localized population distribution and anecdotal information from naturalist observations, although little information exists on dispersal ability and distances (Burke et al. 2011). Predicted habitat requirements based on field observations outlined in the recovery report for the species includes presence of the host plant(s) (Olson 2002) in multiple patches (Schweitzer et al. 2011), presence of nectar plants (COSEWIC 2015), partial shade (Olson 2002), moist soil (Schweitzer et al. 2011), and

variable land topography (Scott 1986). Mottled Duskywing are opportunistic generalists regarding nectar plant preferences and have been observed in the field to be frequently nectaring on their larval host plants, as well as Oxeye Daisy, Red Clover, Canada Anemone, Wild Columbine and more (Natural Resource Solutions Incorporated, unpublished data).



Figure 1-1. Photo of a Mottled Duskywing (*Erynnis martialis*) in the wild. Photo credit: Shayla Kroeze.



Figure 1-2. Mottled Duskywing host plants, New Jersey Tea (Figure A; *Ceanothus americanus*; photo credit: Shayla Kroeze) and Prairie Red Root (Figure B; *Ceanothus herbaceus*; photo credit: Bob Bell).

The Mottled Duskywing was listed as endangered by the Committee on the Status of Endangered Wildlife in Canada (COSEWIC) in 2012 (COSEWIC 2015) and is the only species of butterfly listed as endangered by the Committee on the Status of Species at Risk in Ontario (COSSARO) under the Ontario Endangered Species Act (ESA; Ontario Provincial Government 2022). It is restricted to a few small, isolated subpopulations where its larval host plants occur (COSEWIC 2015; Figure 1-3). Historically in Canada, the Mottled Duskywing was found in southeastern Manitoba, southern Ontario, and southwestern Quebec. The species was last observed in Quebec in the 1950s and is thought to be extirpated from the province (COSEWIC 2012). Additionally, severe population declines have occurred in the United States (COSEWIC 2015).

Currently in Ontario, known persisting populations are in Burlington, Roseneath, Centreton, and Marmora (Figure 1-3). All sites where the species is found are variable in their environmental topography and characteristics. In addition, small numbers of Mottled Duskywing have been observed in Camp Borden, Oakville, and other locations in the Rice Lake Plains. Threats to the Mottled Duskywing include loss of habitat, pesticide use, host plant competition with invasive species, and climate change (COSEWIC 2015). Limitations to recovery include the Mottled Duskywing's high degree of habitat specificity, predicted low dispersal ability (COSEWIC 2015), and shifting phenology caused by climate change (Kingsford and Watson 2011).


Figure 1-3. Current and historical distribution of the Mottled Duskywing (*Erynnis martialis*) in Ontario. Originally published in COSEWIC 2015, updated in 2016 by Natural Resources Solutions Incorporated. Permission to include from Jessica Linton (NRSI).

1.4.1 Reintroduction efforts for Mottled Duskywing

As part of the recommended recovery plan, a collaborative effort began in 2017 to release Mottled Duskywing individuals in Pinery Provincial Park, where the species historically occurred. That reintroduction has been initiated, with the first founders having been introduced into Pinery in the summer of 2021, and was associated with a mark-recapture program, captive breeding, and this genetic research. This work was authorized by the Ministry of Environment, Conservation and Parks through a permit under the Endangered Species Act. This first reintroduction site may eventually be followed by reintroduction to other locations in Ontario, including one tentatively planned in Norfolk County, as well as augmentation of currently extant but small populations. Conservation of the Mottled Duskywing is not only beneficial for the species, but also promotes protection of globally rare oak-savanna ecosystems, as well as tallgrass prairie communities, and the many species that utilize these habitat types at Pinery Provincial Park (Marotta 2021).

1.5 Thesis objectives

My research aimed to support and inform conservation reintroduction efforts for the Mottled Duskywing by:

- 1. Developing microsatellite loci for *Erynnis martialis* that can be used for populations genetic analyses (Chapter 2).
- 2. Characterizing genetic diversity among extant populations of Mottled Duskywing in Ontario, as well as known populations in Manitoba, New York, and Michigan

to assess if populations are sufficiently diverse to support a reintroduction (Chapter 3).

- 3. Characterizing genetic differentiation among populations to better understand pattens of gene flow and connectivity, provide insight into potential dispersal ability, and assess the potential mixing of populations for reintroduction sourcing (Chapter 3).
- Estimating effective population sizes and testing for evidence of recent bottlenecks in extant populations to provide new insight into demographic stability (Chapter 3).

1.6 References

- Andersen A, Simcox DJ, Thomas JA, Nash DR (2014). Assessing reintroduction schemes by comparing genetic diversity of reintroduced and source populations: A case study of the globally threatened large blue butterfly (*Maculinea arion*). *Biological Conservation* 175: 34–41. doi: 10.1016/j.biocon.2014.04.009.
- Andrews TM, Price RM, Mead LS, et al. (2017). Biology undergraduates' misconceptions about genetic drift. CBE–Life Sciences Education 11: 3. doi: 10.1187/cbe.11-12-0107.
- Anthony N, Gelembiuk G, Raterman D, Nice C, Ffrench-Constant R (2001). Isolation and characterization of microsatellite markers from the endangered Karner Blue butterfly *Lycaeides melissa samuelis* (Lepidoptera). *Hereditas* 134: 271–273. doi: 10.1111/j.1601-5223.2001.00271.x.
- Betrus CJ, Fleishman E, Blair RB (2005). Cross-taxonomic potential and spatial transferability of an umbrella species index. *Journal of Environmental Management* 74: 79–87. doi: 10.1016/j.jenvman.2004.08.010.
- Bhargava A, Fuentes FF (2010). Mutational dynamics of microsatellites. *Molecular Biotechnology* 44: 250–266. doi: 10.1007/s12033-009-9230-4.
- Brook BW, Sodhi NS, Bradshaw CJA (2008). Synergies among extinction drivers under global change. *Trends in Ecology and Evolution* 23: 453–460. doi: 10.1016/j.tree.2008.03.011.
- Broquet T, Petit EJ (2009). Molecular estimation of dispersal for ecology and population genetics. Annual Review of Ecology, Evolution, and Systemics 40: 193–216. doi: 10.1146/annurev.ecolsys.110308.120324.
- Brown CJ, Broadley A, Adame MF, Branch TA, Turschwell MP, Connolly RM (2018). The assessment of fishery status depends on fish habitats. *Fish and Fisheries* 20: 1–14. doi: 10.1111/faf.12318.
- Brown TA (2002). Mutation, Repair, and Recombination. Brown TA (Ed.). Genomes 2nd Edition. Oxford, New York.
- Burke RJ, Fitzsimmons JM, Kerr JT (2011). A mobility index for Canadian butterfly species based on naturalists' knowledge. *Biodiversity and Conservation* 20: 2273–2295. doi: 10.1007/s10531-011-0088-y.
- Burkey TV and Reed DH (2006). The effects of habitat fragmentation on extinction risk: Mechanisms and synthesis. *Songklanakarin Journal of Science and Technology* 28: 9–37.
- Ceballos G, Ehrlich PR, Barnosky AD, García A, Pringle RM, Palmer TM (2015). Accelerated modern human-induced species losses: Entering the sixth mass extinction. *Science Advances* 1: e1400253. doi: 10.1126/sciadv.1400253.
- Charlesworth B (2009). Effective population size and patterns of molecular evolution and variation. *Nature Reviews Genetics* 10: 195–205. doi: 10.1038/nrg2526.
- Charlesworth B, Charlesworth D (1999). The genetic basis of inbreeding depression. *Genetical Research* 74: 329–340. doi: 10.1038/nrg2664.
- Charlesworth D, Charlesworth B (1987). Inbreeding depression and its evolutionary consequences. *Annual Review of Ecology and Systematics* 18(1): 237–268.

- Chistiakov DA, Hellemans B, Volckaert FAM (2006). Microsatellites and their genomic distribution, evolution, function and applications: A review with special reference to fish genetics. *Aquaculture* 255: 1–29. doi: 10.1016/j.aquaculture.2005.11.031.
- COSEWIC (2012). COSEWIC Assessment and Status Report on the Mottled Duskywing *Erynnis martialis* in Canada. COSEWIC Committee on the Status of Endangered Wildlife in Canada. Retrieved January 19, 2022, from https://www.registrelepsararegistry.gc.ca/virtual_sara/files/cosewic/sr_hesperie_tachetee_mottled_dusky wing_1213_e.pdf.
- COSEWIC (2015). Recovery Strategy for the Mottled Duskywing (*Erynnis martialis*) in Ontario. Canada: Ontario Ministry of Natural Resources and Forestry. Ontario Recovery Strategy Series. Retrieved March 27, 2020, from https://collections.ola.org/mon/29008/331486.pdf.
- Daniels JC, Nordmeyer C, Runquist E (2018). Improving standards for at-risk butterfly translocations. *Diversity* 10: 67–80. doi: 10.3390/d10030067.
- De Barba M, Waits LP, Garton EO, Genovesi P, Randi E, Mustoni A, Groffs C (2010). The power of genetic monitoring for studying a demography, ecology and genetics of a reintroduced brown bear population. *Molecular Ecology* 19: 3938– 3951. doi: 10.1111/j.1365-294X.2010.04791.x.
- Deere JA, Coulson T, Cubaynes S, Smallegange IM (2017). Unsuccessful dispersal affects life history characteristics of natal populations: The role of dispersal related variation in vital rates. *Ecological Modelling* 366: 37–47. doi: 10.1016/j.ecolmodel.2017.10.010.
- Díaz S, Fargione J, Chapin III FS, Tilman D (2006). Biodiversity loss threatens human well-being. *Public Library of Science Biology* 4(8): e277. doi: 10.1371/journal.pbio.0040277.
- Dirzo R, Young HS, Galetti M, Ceballos G, Isaac NJB, Collen B (2014). Defaunation in the Anthropocene. *Science* 345(6195): 401–406. doi: 10.1126/science.1251817.
- Dumeier AC, Lorenz AW, Kiel E (2020). Active reintroduction of benthic invertebrates to increase stream biodiversity. *Limologica* 80: 125726. doi: 10.1016/j.limno.2019.125726.
- Duminil J, Fineschi S, Hampe A, Jordano P, Salvini D, Vendramin GG, Petit RJ (2007). Can population genetic structure be predicted from life history traits? *The American Naturalist* 169: 662–672. doi: 10.1086/513490.
- Dunn RR (2005). Modern insect extinctions, the neglected majority. *Conservation Biology* 19: 1030–1036. doi: 10.1111/j.1523-1739.2005.00078.x.
- Dyer JM (1995). Assessment of climatic warming using a model of forest species migration. *Ecological Modelling* 79: 199–219. doi: 10.1016/0304-3800(94)00038-J.
- Edmands S (2007). Between a rock and a hard place: evaluating the relative risks of inbreeding and outbreeding for conservation and management. *Molecular Ecology* 16: 463–475. doi: 10.1111/j.1365-294X.2006.03148.x.
- Ehlers A, Worm B, Reusch TBH (2008). Importance of genetic diversity in eelgrass *Zostera marina* for its resilience to global warming. *Marine Ecology Progress Series* 355: 1-7. doi: 10.3354/meps07369.
- Ehrlich PR (1992). Population biology of checkerspot butterflies and preservation of global diversity. *Oikos* 63: 6–12.

- Eisenhauer N, Bonn A, Guerra CA (2019). Recognizing the quiet extinction of invertebrates. *Nature Communications* 10: 50. doi: 10.1038/s41467-018-07916-1.
- Ellegren H (2004). Microsatellites: simple sequences with complex evolution. *Nature Reviews Genetics* 5: 435–445. doi: 10.1038/nrg1348.
- Feeley KJ, Rehm EM, Machovina B (2012). The responses of tropical forest species to global climate change: acclimate, adapt, migrate, or go extinct? *Frontiers of Biogeography* 4: 2. doi: 10.21425/F5FBG12621.
- Fleishman E, Thomson JR, Mac Nally R, Murphy DD, Fay JP (2005). Using indicator species to predict species richness of multiple taxonomic groups. *Conservation Biology* 19: 1125–1137. doi: 10.1111/j.1523-1739.2005.00168.x.
- Franke K, Fischer K (2013). Effects of inbreeding and temperature stress on life history and immune function in a butterfly. *Journal of Evolutionary Biology* 26: 517–528. doi: 10.1111/jeb.12064.
- Frankham R, Ballou JD, Eldridge MD, Lacy RC, Ralls K, Dudash MR, Fenster CB (2011). Predicting the probability of outbreeding depression. *Conservation Biology* 25: 465–475. doi: 10.1111/j.1523-1739.2011.01662.x.
- Frankham R., Ballou JD, Ralls K, Eldridge M, Dudash MR, Fenster CB, et al. (2017). Genetic management of fragmented animal and plant populations. Oxford University Press, Oxford.
- Frankham R, Ralls K (1998). Inbreeding leads to extinction. *Nature* 392: 441–442. doi: 10.1038/33022.
- Fraser DJ (2008). How well can captive breeding programs conserve biodiversity? A review of salmonids. *Evolutionary Applications* 1: 535–586. doi: 10.1111/j.1752-4571.2008.00036.x.
- Geyle HM, Braby MF, Andren M, et al. (2021). Butterflies on the brink: identifying the Australian butterflies (Lepidoptera) most at risk of extinction. *Austral Entomology* 60: 98–110. doi: 10.1111/aen.12525.
- Gregory TR (2009). Understanding natural selection: essential concepts and common misconceptions. *Evolution: Education and Outreach* 2: 156–175. doi: 10.1007/s12052-009-0128-1.
- Gusset M (2009). A Framework for Evaluating Reintroduction Success in Carnivores: Lessons from African Wild Dogs. Hayward MW, Somers M (Eds.). Reintroduction of Top-Order Predators. Wiley-Blackwell, New York.
- Haikola S, Fortelius W, O'Hara RB, Kuussaari M, Wahlberg N, Saccheri IJ, Singer MC, Hanski I (2001). Inbreeding depression and the maintenance of genetic load in *Melitaea cinxia* metapopulations. *Conservation Genetics* 2: 325–335. doi: 10.1023/A:1012538329691.
- Hansen MM, Kenchington E, Nielsen EE (2001). Assigning individual fish to populations using microsatellite DNA markers. *Fish and Fisheries* 2: 93–112. doi: 10.1046/j.1467-2960.2001.00043.x.
- Hanski I (2005). Landscape fragmentation, biodiversity loss and the societal response. *MBO Reports* 6: 388–392. doi: 10.1038/sj.embor.7400398.
- Hanski I, Pakkala T, Kuussaari M, Lei G (1995). Metapopulation persistence of endangered butterfly in fragmented landscape. *Oikos* 72: 21–28. doi: 10.2307/3546033.

- Hansson B, Bensch S, Hasselquist D, Åkesson M (2001) Microsatellite diversity predicts recruitment of sibling great reed warblers. *Proceedings of the Royal Society Biological Sciences* 268: 1287–1291. doi: 10.1098/rspb.2001.1640.
- Hedrick PW (2001). Conservation genetics: where are we now? *Trends in Ecology and Evolution* 16: 629–636. doi: 10.1016/S0169-5347(01)02282-0.
- Hedrick PW (2011). Genetics of Populations. Jones and Bartlett Learning, Sudbury.
- Hedrick PW, Garcia-Dorado A (2016). Understanding inbreeding depression, purging, and genetic rescue. *Trends in Ecology and Evolution* 31: 940–952. doi: 10.1016/j.tree.2016.09.005.
- Hedrick PW, Kalinowski ST (2000). Inbreeding depression in conservation biology. *Annual Review of Ecology and Systematics* 31: 139–162. doi: 10.1146/annurev.ecolsys.31.1.139.
- Hendry AP, Taylor EB, McPhail JD (2002). Adaptive divergence and the balance between selection and gene flow: lake and stream stickleback in the Misty system. *Evolution* 56: 1199–1216. doi: 10.1111/j.0014-3820.2002.tb01432.x.
- Houde ALS, Garner SR, Neff BR (2015). Restoring species through reintroductions: strategies for source population selection. *Restoration Ecology* 23: 746–753. doi: 10.1111/rec.12280.
- Hughes AR, Stachowicz JJ (2004). Genetic diversity enhances the resistance of a seagrass ecosystem to disturbance. *Proceedings of the National Academy of Sciences* 101(24): 8998–9002. doi: 10.1073/pnas.0402642101.
- Hughes JM, Stewart J, Lyle JM, McAllister J, Stocks JR, Suthers IM (2017). Influence of latitudinal variation in environmental gradients and population structure on the demography of a widespread pelagic fish, *Arripis trutta* (Forster, 1801). *Environmental Biology of Fishes* 100: 121–135. doi: 10.1007/s10641-016-0565-y.
- Husemann M, Zachos FE, Paxton RJ, Habel JC (2016). Effective population size in ecology and evolution. *Heredity* 117: 191–192. doi: 10.1038/hdy.2016.75.
- IPCC (2014). RK Pachauri, LA Meyer (Eds.) Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (p. 151). Core Writing Team, IPCC, Geneva, Switzerland.
- Isaak DJ, Thurow RF, Rieman BE and Dunham, JB (2007). Chinook Salmon use of spawning patches: relative roles of habitat quality, size, and connectivity. *Ecological Applications* 17: 352–364. doi: 10.1890/05-1949.
- IUCN (1987). Translocation of Living Organisms: Introductions, Re-introductions, and Re-stocking. IUCN position statement. Gland, Switzerland.
- IUCN (1998). IUCN Guidelines for Re-introductions. IUCN, Gland, Switzerland.
- IUCN/SSC (2013). Guidelines for Reintroductions and Other Conservation Translocations. Version 1.0 IUCN Species Survival Commission, Gland, Switzerland viiii + 57 pp. Retrieved February 2, 2022, from https://portals.iucn.org/library/sites/library/files/documents/2017-065.pdf.
- Jetz W, Wilcove DS, Dobson AP (2007). Projected impacts of climate and land-use change on the global diversity of birds. *PLoS Biology* 5: 1211–1219. doi: 10.1371/journal.pbio.0050157.

- Jourdan J, Plath M, Tonkin JD, et al. (2019). Reintroduction of freshwater macroinvertebrates: challenges and opportunities. *Biological Reviews* 94: 368– 387. doi: 10.1111/brv.12458.
- Jule KR, Leaver LA, Stephen EG (2008). The effects of captive experience on reintroduction survival in carnivores: A review and analysis. *Biological Conservation* 141: 355–363. doi: 10.1016/j.biocon.2007.11.007.
- Kelemen EP, Rehan SM (2021). Conservation insights from wild bee genetic studies: Geographic differences, susceptibility to inbreeding, and signs of local adaptation. *Evolutionary Applications* 14: 1485–1496. doi: 10.1111/eva.13221.
- Kingsford RT, Watson JE (2011). Climate change in Oceania: a synthesis of biodiversity impacts and adaptations. *Pacific Conservation Biology* 17: 270–284. doi: 10.1071/PC110270.
- Kimura M, Crow JF (1963), The measurement of effective population number. *Evolution* 17: 279–288. doi: 10.1111/j.1558-5646.1963.tb03281.x.
- Kronenberger JA, Gerberich JC, Fitzpatrick SW, Broder ED, Angeloni LM, Funk WC (2018). An experimental test of alternative population augmentation scenarios. *Conservation Biology* 32: 838–848. doi: 10.1111/cobi.13076.
- Lande R, Shannon S (1996). The role of genetic variation in adaptation and population persistence in a changing environment. *Evolution* 50: 434–437. doi: 10.1111/j.1558-5646.1996.tb04504.x.
- Lebreton J, Burnham KP, Clobert J, Anderson DR (1992). Modeling survival and testing biological hypotheses using marked animals: a unified approach with case studies. *Ecological Monographs* 62: 67–118. doi: 10.2307/2937171/
- Lemoine N, Bauer H-G, Peintinger M, Böhning-Gaese K (2007). Effects of climate and land-use change on species abundance in a Central European bird community. *Conservation Biology* 21: 495–503. doi: 10.1111/j.1523-1739.2006.00633.x.
- Lesica P, Allendorf FW (1999). Ecological genetics and the restoration of plant communities: mix or match? *Restoration Ecology* 7: 42–50. doi: 10.1046/j.1526-100X.1999.07105.x.
- Loewe L, Hill WG (2010). The population genetics of mutations: good, bad and indifferent. *Philosophical Transactions of the Royal Society Biology* 365: 1153–1167. doi: 10.1098/rstb.2009.0317.
- Losey JE, Vaughan M (2006). The economic value of ecological services provided by insects. *BioScience* 56(4): 311–323. doi: 10.1641/0006-3568.
- Lowe WH, Kovach RP, Allendorf FW (2017). Population genetics and demography unite ecology and evolution. *Trends in Ecology and Evolution* 32: 141–152. doi: 10.1016/j.tree.2016.12.002.
- Luikart G, Sherwin WB, Steele BM, Allendorf FW (1998). Usefulness of molecular markers for detecting population bottlenecks via monitoring genetic change. *Molecular Ecology* 7: 963–974. doi: 10.1046/j.1365-294x.1998.00414.x.
- Lynch M (1991). The genetic interpretation of inbreeding depression and outbreeding depression. *Evolution* 45: 622–629. doi: 10.1111/j.1558-5646.1991.tb04333.x.
- Mantya-pringle CS, Martin TG, Rhodes JR (2012). Interactions between climate and habitat loss effects on biodiversity: a systematic review and meta-analysis. *Global Change Biology* 18: 1239–1252. doi: 10.1111/j.1365-2486.2011.02593.x.

- Marotta S (2021). Endangered mottled duskywing to be reintroduced in Ontario. The Globe and Mail, Toronto, Ontario. Retrieved February 20, 2022, from https://www.theglobeandmail.com/canada/article-endangered-mottledduskywing-to-be-reintroduced-inontario/#:~:text=Linton%20as%20chair%20of%20the,of%20an%20endangered%
- 20butterfly%20species. Maruyama T, Fuerst PA (1985). Population bottlenecks and nonequilibrium models in populations genetics. III. Genic homozygosity in populations which experience periodic bottlenecks. *Genetics* 111: 697–703. doi: 10.1093/genetics/111.3.691.
- Mayr E (1963). Animal species and evolution. Harvard University Press, Cambridge, Mass. pp. 211–212.
- McClelland E, Naish K (2007). What is the fitness outcome of crossing unrelated fish populations? A meta-analysis and an evaluation of future research directions. *Conservation Genetics* 8: 397–416. doi: 10.1007/s10592-006-9178-x.
- McDermott JM, McDonald BA (1993). Gene flow in plant pathosystems. *Annual Review* of *Phytopathology* 31: 353–373. doi: 10.1146/annurev.py.31.090193.002033.
- McNeely JA, Miller KR, Reid WV, Mittermeier RA, Werner TB (1990). Conserving the world's biological diversity. IUCN, World Resources Institute, Conservation International, WWF-US and the World Bank: Washington, DC.
- Miles LS, Rivkin LR, Johnson MTJ, Munshi-South J, Verrelli BC (2019). Gene flow and genetic drift in urban environments. *Molecular Ecology* 28: 4138–4151. doi: 10.1111/mec.15221.
- Munshi-South J, Zolnik CP, Harris SE (2016). Population genomics of the Anthropocene: Urbanization is negatively associated with genome-wide variation in white-footed mice populations. *Evolutionary Applications* 9: 546–564. doi: 10.1111/eva.12357.
- O'Brien SJ, Evermann JF (1988) Interactive influence of infectious disease and genetic diversity in natural populations. *Trends in Ecology and Evolution* 3: 254–259. doi: 10.1016/0169-5347(88)90058-4.
- Oliver TH, Morecroft MD (2014). Interactions between climate change and land use change on biodiversity: attribution problems, risks, and opportunities. *WIREs Climate Change* 5: 317–335. doi: 10.1002/wcc.271.
- Olson S (2002). Conservation Assessment for Mottled Duskywing (*Erynnis martialis*). USDA Forest Service, Eastern Region pp 1–9.
- Ontario Provincial Government (2022). Mottled Duskywing, Ministry of the Environment, Conservation and Parks. Retrieved February 22, 2022, from https://www.ontario.ca/page/mottledduskywing#:~:text=Where%20it's%20been%20found%20in,Marmora%20(east% 20of%20Peterborough).
- Palstra FP, Ruzzante DE (2008). Genetic estimates of contemporary effective population size: what can they tell us about the importance of genetic stochasticity for wild population persistence? *Molecular Ecology* 17: 3428–3447. doi: 10.1111/j.1365-294X.2008.03842.x.
- Parmesan C (2006). Ecological and evolutionary responses to recent climate change. Annual *Review of Ecology, Evolution, and Systematics* 37: 637–669. doi: 10.1146/annurev.ecolsys.37.091305.110100.

- Pe'er G, Settele J (2008). Butterflies in and for conservation: trends and prospects. *Israel Journal of Ecology and Evolution* 54: 7–17. doi: 10.1560/ijee.54.1.7.
- Pievani T (2014). The sixth mass extinction: Anthropocene and the human impact on biodiversity. *Rendiconti Lincei-Scienze Fishe e Naturali* 25: 85–93. doi: 10.1007/s12210-013-0258-9.
- Pimm SL (2008). Biodiversity: Climate change or habitat loss which will kill more species? *Current Biology* 18: R117–R119. doi: 10.1016/j.cub.2007.11.055.
- Plutynski A (2007). Neutralism. Matthen M, Stepends C (Eds.). Handbook of the Philosophy of Science (pp. 129–140). Philosophy of Biology, North-Holland. doi: 10.1016/B978-044451543-8/50009-5.
- Prather CM, Pelini SL, Laws A, et al. (2013). Invertebrates, ecosystem services and climate change. *Biological Reviews* 88: 327–348. doi: 10.1111/brv.12002.
- Quintero I, Wiens JJ (2013). Rates of projected climate change dramatically exceed past rates of climatic niche evolution among vertebrate species. *Ecology Letters* 16: 1095–1103. doi: 10.1111/ele.12144.
- Rao VR, Hodgkin T (2002). Genetic diversity and conservation and utilization of plant genetic resources. *Plant Cell, Tissue and Organ Culture* 68: 1–19. doi: 10.1023/A:1013359015812.
- Saccheri IJ, Brakefield PM, Nichols RA (1996). Severe inbreeding depression and rapid fitness rebound in the butterfly *Bicyclus anynana* (Satyridae). *Evolution* 50: 2000–2013. doi: 10.1111/j.1558-5646.1996.tb03587.x.
- Saccheri I, Kuussaari M, Kankare M, Vikman P, Fortelius W, Hanski I (1998). Inbreeding and extinction in a butterfly metapopulation. *Nature* 392: 491–494. doi: 10.1038/33136.
- Sah P, Mandal S, Singh RK et al. (2021) Development of novel microsatellite marker panel in threatened tetraploid mahseer, *Tor tor* (Hamilton 1822) for insights into its genetic diversity and population structure. *Meta Gene* 28: 100880. doi: 10.1016/j.mgene.2021.100880.
- Sánchez-Bayo F, Wyckhuys KAG (2019). Worldwide decline of the entomofauna: A review of its drivers. *Biological Conservation* 232: 8–27. doi: 10.1016/j.biocon.2019.01.020.
- Savolainen O, Pyhäjärvi T, Knürr T (2007). Gene flow and local adaptation in trees. Annual Review of Ecology, Evolution, and Systematics 38: 595–619. doi: 10.1146/annurev.ecolsys.38.091206.095646.
- Schmitt T, Hewitt GM (2004). The genetic pattern of population threat and loss: a case study of butterflies. *Molecular Ecology* 13: 21–31. doi: 10.1046/j.1365-294X.2004.02020.x.
- Schultz CB, Russell C, Wynn L (2008). Restoration, reintroduction, and captive propagation for at-risk butterflies: A review of British and American conservation efforts. *Israel Journal of Ecology and Evolution* 54: 41–61. doi: 10.1560/IJEE.54.1.41
- Schweitzer DF, Minno MC, Wagner DL (2011). Rare, Declining, and Poorly Known Butterflies and Moths (Lepidoptera) of Forests and Woodlands in the Eastern United Stations. U.S. Forest Service. Forest Health Technology Enterprise Team, United States of America.

- Scott JA (1986). The Butterflies of North America: A Natural History and Field Guide. Stanford University Press, Stanford, California.
- Seddon PJ (1999). Persistence without intervention: Assessing success in wildlife reintroductions. *Trends in Ecology and Evolution* 14: 503. doi: 10.1016/S0169-5347(99)01720-6.
- Sexton JP, Hangartner SB, Hoffmann AA (2014). Genetic isolation by environment or distance: which pattern of gene flow is most common? *Evolution* 68: 1–15. doi: 10.1111/evo.12258.
- Shen Y, Wang L, Fu J, Xu X, Yue GH, Li J (2019). Population structure, demographic history and local adaptation of the grass carp. *BMC Genomics* 20: 467. doi: 10.1186/s12864-019-5872-1.
- Short J, Bradshaw SD, Giles J, Prince RIT, Wilson GR (1992). Reintroduction of macropods (Marsupialia: Macropodoidea) in Australia—A review. *Biological Conservation* 62: 189–204. doi: 10.1016/0006-3207(92)91047-V.
- Slatkin M (1985). Gene flow in natural populations. *Annual Review of Ecology, Evolution, and Systematics* 16: 393–430, doi: 10.1146/annurev.es.16.110185.002141.
- Spangenberg JH, Settele J (2010). Precisely incorrect? Monetising the value of ecosystem services. *Ecological Complexity* 7: 327–337. doi: 10.1016/j.ecocom.2010.04.007.
- Star B, Spencer HG (2013). Effects of genetic drift and gene flow on the selective maintenance of genetic variation, *Genetics* 194: 235–244. doi: 10.1534/genetics.113.149781.
- Stork NE (2018). How many species of insects and other terrestrial arthropods are there on Earth? *Annual Review of Entomology* 63: 31–45. doi: 10.1146/annurev-ento-020117-043348.
- Tarsi K, Tuff TR (2012) Introduction to population demographics. *Nature Education Knowledge* 3: 3.
- Templeton AR (1980). The theory of speciation *via* the founder principle. *Genetics* 94: 1011–1038. doi: 10.1093/genetics/94.4.1011.
- Thomas CD, Franco AMA, Hill JK (2006). Range retractions and extinction in the face of climate warming. *Trends in Ecology and Evolution* 21: 415–416. doi: 10.1016/j.tree.2006.05.012.
- Tracy LN, Wallis GP, Efford MG, Jamieson IG (2011). Preserving genetic diversity in threatened species reintroductions: how many individuals should be released? *Animal Conservation* 14: 439–446. doi: 10.1111/j.1469-1795.2011.00448.x.
- Uehara-Prado M, Freitas AV (2009). The effect of rainforest fragmentation on species diversity and mimicry ring composition of ithomiine butterflies. *Insect Conservation and Diversity* 2: 23–28. doi: 10.1111/j.1752-4598.2008.00025.x.
- van der Valk T, de Manuel M, Marques-Bonet T, Guschanski K (2021). Estimates of genetic load suggest frequent purging of deleterious alleles in small populations. *bioRxiv*: 696831. doi: 10.1101/696831.
- Vickery M (2008). Butterflies as indicators of climate change. *Science Progress* 91: 193–201. doi: 10.3184/003685008X327927.
- Vieira ML, Santini L, Diniz AL, Munhoz Cde F (2016). Microsatellite markers: what they mean and why they are so useful. *Genetics and Molecular Biology* 39: 312–328. doi: 10.1590/1678-4685-GMB-2016-0027.

- Wagner DL, Van Driesche RG (2010). Threats posed to rare or endangered insects by invasions of nonnative species. *Annual Review of Entomology* 55: 547–568. doi: 10.1146/annurev-ento-112408-085516.
- Waldron A, Miller D, Redding D et al. (2017). Reductions in global biodiversity loss predicted from conservation spending. *Nature* 551: 364–367. doi: 10.1038/nature24295.
- Wang IJ, Bradburd GS (2014). Isolation by environment. *Molecular Ecology* 23: 5649–5662. doi: 10.1111/mec.12938.
- Wang IJ, Glor RE, Losos JB (2013). Quantifying the roles of ecology and geography in spatial genetic divergence. *Ecology Letters* 16: 175–182. doi: 10.1111/ele.12025.
- Warren MS, Maes D, van Swaay CAM, Goffart P, Van Dyck H, Bourn NAD, Wynhoff I, Hoare D, Ellis S (2021). The decline of butterflies in Europe: Problems, significance, and possible solutions. *Proceedings of the National Academy of Sciences* 118(2): e2002551117. doi: 10.1073/pnas.2002551117.
- Weeks AR, Heinze D, Perrin L, Stoklosa J, Hoffmann AA, van Rooyen A, et al. (2017). Genetic rescue increases fitness and aids rapid recovery of an endangered marsupial population. *Nature Communications* 8: 1071. doi: 10.1038/s41467-017-01182-3.
- Wilson EO (1987). The little things that run the world (the importance and conservation of invertebrates). *Conservation Biology* 1: 344–346. doi: 10.1111/j.1523-1739.1987.tb00055.x.
- Wright S (1931). Evolution in Mendelian populations. *Genetics* 16: 97–159. doi: 10.1093/genetics/16.2.97.
- Wright S (1951). The genetical structure of populations. *Annals of Eugenics* 15: 323–354. doi: 10.1111/j.1469-1809.1949.tb02451.x.
- Xu B, Sun G, Wang X, Lu J, Wang IJ, Wang Z (2017). Population genetic structure is shaped by historical, geographic, and environmental factors in the leguminous shrub *Caragana microphylla* on the Inner Mongolia Plateau of China. *BMC Plant Biology* 17: 200. doi: 10.1186/s12870-017-1147-7.

Chapter 2

2 Development of tetranucleotide and dinucleotide microsatellites for the at-risk Mottled Duskywing butterfly, *Erynnis martialis*

2.1 Abstract

Twenty-four microsatellite loci for an at-risk North American butterfly, the Mottled Duskywing (*Erynnis martialis*), were isolated and characterized. Forty-one tetraand di-nucleotide loci were originally identified and selected using Illumina nextgeneration sequencing, and these were refined to 24 variable and cleanly amplifiable loci using fragment analysis. I also describe conditions by which these microsatellites can be multiplexed in eight separate reactions. These microsatellite loci were developed to investigate population differentiation and diversity of the Mottled Duskywing in Ontario, Canada where the species is listed as endangered. The microsatellites will be used to inform reintroduction and conservation protocols for the species in Ontario. The number of observed alleles at the 24 microsatellites in a sample of 34 Mottled Duskywing from a single location ranged from 2 to 10 with observed levels of heterozygosity ranging from 0.06 to 0.76 and expected levels of heterozygosity ranging from 0.06 to 0.79.

2.2 Body

Erynnis martialis (Scudder, 1870), commonly known as the Mottled Duskywing, is a medium-sized, brown coloured butterfly belonging to the skipper family (Lepidoptera: *Hesperiidae*; *Pyrginae*; COSEWIC 2015). It is endemic to North America, and due to threats, such as loss of habitat, pesticide use, host plant competition with invasive species, and climate change, in the province of Ontario, Canada the species is now only represented by a limited number of small, isolated populations (COSEWIC 2015). It is listed as endangered in this province by the Committee on the Status of Endangered Wildlife in Canada. The only previously published genetic information on the Mottled Duskywing consists of DNA barcodes (COSEWIC 2012).

Microsatellites were developed to characterize genetic diversity and differentiation of extant populations and inform planned reintroductions of the species to locations from which it has been extirpated. Previous research shows that it is critical to consider genetic structure and diversity of a species to optimize strategies for reintroduction programs (Daniels et al. 2018), particularly for selection of a source population (Drauch et al. 2008). Microsatellites were selected because of their codominance and high variability (Regnaut et al. 2006; Chassaing et al. 2018), particularly given that a limited number of individual butterflies were initially available for marker development (12 female butterflies from a single population, with a 2019 population size estimate of 252-513 individuals; Demarse, unpublished data).

Genomic DNA was extracted using a DNeasy Blood and Tissue Kit (Qiagen, Germantown, MD) from 12 whole, mated female butterflies provided by a simultaneous captive rearing program located at the Cambridge Butterfly Conservatory (Ontario, Canada). A sterile pestle was used to manually crush and grind tissue from the head, thorax, and legs of each individual; the abdomen was excluded to avoid risk of including DNA from sperm. A single elution with 200 µl of MilliQ water warmed to 36 °C was used to collect the DNA and a Qubit fluorometer (Invitrogen) was used to quantify the DNA. An equimolar DNA pool from all twelve individuals (105.84 µL total volume) was sent to GenoScreen Services (Lille, France; www.genoscreen.fr), where the DNA pool was enriched for microsatellite loci and subsequently sequenced using their Geno Sat® service. In brief, a microsatellite-enriched genomic library was developed and sequenced using the Illumina MiSeq platform to generate 250 bp paired-end reads following manufacturer guidelines from MiSeq Reagent Nano Kit v2. Approximately 1.30 million reads were obtained and merged into 590,832 contigs with PrinSeq software. The bioinformatics program QDD v3 (Meglécz et al. 2014) was used to detect and select microsatellites and design primers. A total of 836 primer pairs were returned and of these, I tested 41 for polymorphism and consistency in scoring. Only perfect di- and tetranucleotide repeats were considered to avoid potential introduction of stop codons within reading frames, and to increase the probability of accurate scoring.

Twenty-two additional Mottled Duskywings were sampled non-lethally in Marmora, Ontario, Canada; this is the same population from which the 12 females used to identify the microsatellites originated. DNA for each of these 22 individuals was extracted from small, non-lethally sampled pieces of wing tissue (Koscinski et al. 2011) using the Qiagen DNeasy Blood and Tissue kit, with a 200 μ L final elution with Milli Q water warmed to 36 °C. Both the non-lethal samples and the whole females were collected in the same year (2019), and the genotype data from all 34 individuals was combined to characterize variation at the microsatellite loci.

Each of the 41 selected di- and tetra-nucleotide microsatellites were initially amplified in a sub-set of two to three individuals in a 10 μ l PCR reaction containing

forward and reverse primers (0.2 μ M each), buffer (1X), MgCl₂ (3 mM), BSA (0.15 mg/ml), dNTPs (0.2 mM), AmpliTaq DNA polymerase (0.25 U), and DNA template. The thermal cycling parameters used for amplification were: 96 °C for 5 min; 30 cycles of denaturation at 96 °C for 30 s, annealing at various temperatures for 30 s, and elongation at 72 °C for 60 s; final extension at 72 °C for 5 min. The optimized annealing temperature varied by locus as designated (Table 2-1).

Table 2-1. Characterization of 24 microsatellite loci in the Mottled Duskywing (*Erynnis martialis*). N_R is the range of the number of repeats, T_m is the annealing temperature in °C, *N* is the number of individuals successfully genotyped (out of 34), bp is the size range of the microsatellite in base pairs, N_a is the number of unique alleles observed, H_o is the

observed heterozygosity	i, and H _E is	s the expected	heterozygosity.
20.	/	1	20 2

Locus	Repeat Unit	Primer sequence (5'–3')	N _R	T _m	Ν	bp	Na	H _O	H _E
EMusat2	ACAT	F: TCAGCTTTATGTCCGGGCAA	39–48	54	29	157-193	4	0.62	0.53
		R: GCCACTACTTGCGACTAATGAG							
EMusat3	AAAT	F: TGGACTGGTGTACCGAAAGT	33-46	54	27	132-184	5	0.47	0.56
		R: TCCTTTCCGTTATATCAAGAAGCG							
EMusat4	AAAG	F: TCAGTAAGGCAAGACTGTTAGCA	64–74	54	34	254-294	9	0.76	0.76
		R: AGTATGGATTAGAGTATCGAGGTCT							
EMusat5	AATT	F: TCTTGATCTAACGTAGTATCGTACG	56–58	53	31	224–232	2	0.62	0.41
		R: TCTACGCAATGCATCAATATTTGA							
EMusat8	AAAC	F: CGCCGGTACTGACAGCTAAA	61-80	55	33	245-321	8	0.62	0.68
		R: GCGAGCGAAATGAGTCCAGA							
EMusat9	ACCT	F: CTGAAGGTGTATCACAGGCAGT	61–68	55	34	243-271	3	0.65	0.62
		R: ACTCTTAACGTCGCGTGTCT					_		
EMusat10	AACG	F: AGGAAATCGTTTGCCAAATACGT	48–52	54	27	192–208	5	0.35	0.56
-		R: TGTACCCTCTATCACAACGAATG	10 10						
EMusat11	AAGG	F: AGACGTATTGGATATTTGGAATGAGT	48–49	54	33	191–195	2	0.21	0.32
EN		R: TGCTAGGACAAGAGCCGATT	44 40	~ 1	24	175 105	~	0.60	0.50
EMusat12	AAAC		44-49	54	34	1/5–195	2	0.62	0.50
EM	ACAT		25 40		22	120 102	2	0.26	0.44
EMUSat15	ACAT		33-49	55	32	139-193	3	0.20	0.44
EMuset14	ACAT		60 72	55	24	275 207	4	0.41	0.50
Elviusat14	ACAI		09-72	55	54	213-201	4	0.41	0.50
EMuset16	ACAC	R. OCCOLAGICAGAGACATAGO	62 66	55	24	247 262	2	0.28	0.28
Elviusatio	ACAG	P. TCATCTCATATCCTTCCTTCCCC	02-00	55	54	247-203	3	0.58	0.38
EMusat17	ΔΔΔΤ		58 50	54	34	221 225	2	0.53	0.50
Liviusat17	АЛЛІ	$\mathbf{R} \cdot \mathbf{GC} \wedge \mathbf{A} \mathbf{TC} \mathbf{GT} \mathbf{T} \wedge \mathbf{A} \wedge \mathbf{C} \mathbf{TT} \mathbf{T} \wedge \mathbf{T} \mathbf{C} \wedge \mathbf{G} \wedge \mathbf{G} \mathbf{G} \mathbf{G} \mathbf{C}$	56-57	54	54	251-255	2	0.55	0.50
EMusat18	AAAC	F. TCCAAACTTGACACCGAGAA	48-60	52	34	191_239	6	0.50	0.59
Linusurio		R. TGGCCTTTATTGTCATTCAATTGT	.0 00	02	5.	171 207	0	0.00	0.07
EMusat20	AAAT	F: TGGCAAAGGTCACTTACGCT	40-44	55	30	160-176	5	0.62	0.56
		R: ACTGGTATGTCCATGGCACC							
EMusat22	AACT	F: ATGGTTCACGGAAGGACCTG	32-39	54	33	126-154	4	0.47	0.59
		R: TGAATACCAAATCGAAGACTGACT							
EMusat25	AAAT	F: CCACGACACAGGGAATCCTA	30-31	54	34	121-125	2	0.26	0.26
		R: GGCTATGACTTCCCGGGTTC							
EMusat27	AAAT	F: CCTGCCATGGGACCAGAAAG	27-28	54	32	107-111	2	0.06	0.06
		R: ACATTTGATGCACCTTATAATATGAG	Г						
EMusat31	AG	F: AGTGTGGATTCAGAAGGCGC	89–97	55	32	178–194	6	0.38	0.41
		R: CACGTTAACTGTCGGGCAAA							
EMusat34	AG	F: GGGTCGGGTCTCCTCATACT	100-114	55	34	200-228	7	0.41	0.56
		R: CCTAGAGTGCACAACTGAGCA							
EMusat35	AC	F: ACTGAGGCCTAGACCCTAGC	112-117	55	33	223-233	5	0.32	0.68
		R: ATTGAGGCCTTCAGACTGCC							
EMusat36	AG	F: GTCCAGTATCTCAGCAGACGG	115-128	55	34	230-256	6	0.44	0.68
		R: AAGGAAGAACAGTGCTAACGCT							
EMusat37	AG	F: ACAGTCTACTAGCCATTCCGT	62–68	54	34	123-135	5	0.56	0.74
		R: TGACCATGACAGCTGTACCAT							
EMusat40	AG	F: AGGCAAATTACAATCAAAGCATCA	139–145	54	34	278-290	6	0.71	0.79
		R: GTGAGCAGTTCGTCGTTTCC							

PCR products were electrophoresed and visualized on agarose gels to check for clean amplification and the presence of bands of the expected sizes. Loci that met these criteria were then amplified using one fluorescently labelled primer per pair (6-FAM, VIC, NED, or PET label), and sized by fragment analysis on a 3730S capillary DNA analyzer, using LIZ-500 size standard (Applied Biosystems, Foster City, CA), at the University of British Columbia's Sequencing and Bioinformatics Consortium (British Columbia, Canada). Genotypes were called using GeneMarker® HID software (SoftGenetics, State College, Pennsylvania, USA). Loci that did not amplify cleanly (i.e., had more than two peaks), or could not be consistently called were removed from further consideration. This left me with 24 loci that I characterized in the full set of 34 individuals. Three loci (EMusat10, EMusat13, EMusat35) showed either significant deviation (P < 0.05) from Hardy Weinberg equilibrium according to analysis using genepop'007 (Rousset 2008) or evidence of null alleles using MICROCHECKER v2.2.3 (Van Oosterhout et al., 2004).

Finally, multiplexes were created to optimize the number of loci that could be simultaneously amplified, by varying the concentrations of primers, as summarized in Table 2-2. I was able to successfully multiplex 18 of the 24 selected loci in six separate reactions.

Table 2-2. Final primer concentrations for PCR amplification of Mottled Duskywing microsatellites including multiplex PCRs. FD refers to the fluorescent dye used to identify each microsatellite, Amp refers to whether the locus amplifies in a multiplex with multiple primers or as a single locus, and Final concentration refers to the final concentration of the corresponding primers in the PCR.

Set	Locus	FD	Amp	Final concentration
Set 1	EMusat14	NED	Multi	0.2 μΜ
	EMusat16	PET	Multi	0.2 μΜ
	EMusat8	VIC	Multi	0.2 μΜ
	EMusat9	6-FAM	Multi	0.2 μΜ
	EMusat25	VIC	Single	0.2 μΜ
Set 2	EMusat36	NED	Multi	0.4 μΜ
	EMusat35	6-FAM	Multi	0.1 μΜ
	EMusat34	VIC	Multi	0.2 μΜ
	EMusat22	NED	Single	0.12 µM
Set 3	EMusat40	NED	Multi	0.2 μΜ
	EMusat4	PET	Multi	1 μ M
	EMusat17	VIC	Multi	0.2 μΜ
	EMusat10	6-FAM	Multi	0.1 μΜ
	EMusat37	NED	Multi	0.2 μΜ
	EMusat27	6-FAM	Multi	0.1 μΜ
Set 4	EMusat18	VIC	Single	0.3 μΜ
Set 5	EMusat31	NED	Multi	0.2 μΜ
	EMusat13	PET	Multi	0.8 μΜ
	EMusat11	6-FAM	Single	0.06 µM
Set 6	EMusat12	NED	Multi	0.2 μΜ
	EMusat2	PET	Multi	0.4 μΜ
	EMusat3	VIC	Multi	0.2 μΜ
Set 7	EMusat20	NED	Single	0.3 μΜ
Set 8	EMusat5	6-FAM	Single	0.3 µM

This set of loci will be a valuable tool for informing conservation of the Mottled Duskywing, including protocols for reintroductions of the species, augmentation of small populations, and assessment and monitoring of genetic diversity and structure in current and introduced populations.

2.3 References

Chassaing O, Desse-Berset N, Hänni C, Hughes S, Berrebi P (2018). Microsatellite diversity of a critically endangered sturgeon, *Acipenser sturio* L. 1758, assessed from museum and archaeological tissue remains. Journal of Biogeography 45: 1043–1053. doi: 10.1111/jbi.13187.

- COSEWIC (2012). COSEWIC Assessment and Status Report on the Mottled Duskywing Erynnis martialis in Canada. COSEWIC Committee on the Status of Endangered Wildlife in Canada. Retrieved January 19, 2022, from https://www.registrelepsararegistry.gc.ca/virtual_sara/files/cosewic/sr_hesperie_tachetee_mottled_dusky wing_1213_e.pdf.
- COSEWIC (2015). Recovery Strategy for the Mottled Duskywing (Erynnis martialis) in Ontario. Canada: Ontario Ministry of Natural Resources and Forestry. Ontario Recovery Strategy Series. Retrieved March 27, 2020, from https://collections.ola.org/mon/29008/331486.pdf.
- Daniels JC, Nordmeyer C, Runquist E (2018). Improving standards for at-risk butterfly translocations. Diversity 10: 67–80. doi: 10.3390/d10030067.
- Drauch AM, Fisher BE, Latch EK, Fike JA, Rhodes OE Jr. (2008). Evaluation of a remnant lake sturgeon population's utility as a source for reintroductions in the Ohio River system. Conservation Genetics 9: 1195–1209. doi: 10.1007/s10592-007-9441-9.
- Koscinski D, Crawford LA, Keller HA, Keyghobadi N (2011). Effects of different methods of non-lethal tissue sampling on butterflies. *Ecological Entomology* 36. doi: 10.1111/j.1365-2311.2011.01272.x.
- Meglécz E, Costedoat C, Dubut V, Gilles A, Malausa T, Pech N, Martin JF (2009). QDD: A user-friendly program to select microsatellite markers and design primers from large sequencing projects. Bioinformatics 26: 403–404. doi: 10.1093/bioinformatics/btp670.
- Regnaut S, Lucas FS, Fumagalli L (2006). DNA degradation in avian faecal samples and feasibility of non-invasive genetic studies of threatened capercaillie populations. Conservation Genetics 7: 449–453. doi: 10.1007/s10592-005-9023-7
- Rousset F (2008). GENEPOP ' 007: a complete re-implementation of the GENEPOP software for Windows and Linux. Molecular Ecology Resources 8: 103–106. doi: 10.1111/j.1471-8286.2007.01931.x.
- Van Oosterhout C, Hutchinson WFD, Wills DP, Shipley P (2004). MICRO-CHECKER: Software for identifying and correcting genotyping errors in microsatellite data. *Molecular Ecology Notes* 4: 535–538. doi: 10.1111/j.1471-8286.2004.00684.x.

Chapter 3

3 Population genetic analysis of the at-risk Mottled Duskywing (*Erynnis martialis*) to inform species reintroductions

3.1 Abstract

Species reintroduction or translocation programs are increasingly used to reverse global declines in biodiversity. Information on genetic diversity and structure within a species can be critical for successful selection of a source population for such programs. The Mottled Duskywing (*Erynnis martialis*) is a North American butterfly listed as endangered in Ontario, Canada, and a collaborative effort to reintroduce the species has begun. Using 21 microsatellite markers, I evaluated metrics of genetic diversity, population structure, and demographic stability of Mottled Duskywing populations in Ontario, Manitoba, and neighbouring U.S. States. These data were used to inform the selection of a source population for an initial reintroduction to Pinery Provincial Park in Grand Bend, Ontario. Mottled Duskywing populations had moderate levels of genetic diversity. Additionally, populations in close proximity had similar allele frequencies, while populations more than 8 km apart had high average genetic differentiation suggesting limited gene flow. Therefore, future management plans might consider populations within an 8 km range as a single management unit. These data can be used to inform future reintroductions or population augmentations of the species, and to assess genetic status of both remnant and reintroduced populations.

3.2 Introduction

Fragmentation, degradation, and loss of suitable habitat coupled with climate change due to growing human populations have caused dramatic loss of global biodiversity (Oliver and Morecroft 2014). It has become necessary to develop effective strategies for the conservation of declining and endangered species (Bainbridge 2014). Several strategies have been implemented to offset this loss of biodiversity including regulations on harmful practices such as overharvesting or release of pollutants, landscape management, invasive species management, and protection of critical habitats (Apollonio et al. 2010; Deinet et al. 2013; Brown 2013). Another strategy that has emerged, often as a final attempt at maintaining or restoring biodiversity in the wild, is species translocation or reintroduction (Seddon et al. 2007; Attard et al. 2016).

The goal of reintroduction is to establish a new population which is selfsustaining and has a low probability of extinction (Fraser 2008). Previous research on reintroduction programs shows that it is critical to consider genetic structure and diversity of a species to optimize strategies for implementation of such programs (Attard et al. 2016, Daniels et al. 2018). Knowledge of the genetic variation within a species is important in conservation for a variety of reasons. Such information helps to define management units and informs the preservation of genetically distinct populations (Haig 1988; Hedrick 2001). Genetic information can provide insights into past changes in population size, connectivity among different populations, potential for inbreeding, and other factors that may contribute to extinction risk through genetic or demographic stochasticity (Yang et al. 2018; Groenen et al. 2012; Lowe and Allendorf 2017). Another important way in which genetic data can inform conservation is through the selection of an appropriate source population or populations for reintroductions (Drauch et al. 2008).

If founders for a new population are chosen from a source population that is small, or too few individuals are used for reintroduction, the new population will have low genetic diversity and a potentially low effective population size (Tracy et al. 2011). Low diversity in the introduced population in turn results in a risk of inbreeding depression and may also lessen the ability of the population to adapt to changing future environmental conditions (Hughes and Sawby 2004). Selecting multiple source populations for reintroduction can increase gene pool variability and improve likelihood of avoiding inbreeding. However, mixing source populations carries its own risks. One such risk is mating incompatibility, which can be caused by pre- or post-zygotic reproductive barriers or by the presence of reproductive parasites such as Wolbachia (Werren et al. 2008). Wolbachia leads to mating barriers through different reproductive phenotypes resulting from infection, the most common being cytoplasmic incompatibility where the sperm and eggs are not able to produce viable offspring if mating individuals do not have the same infection status (Werren et al. 2008). A second risk of mixing source populations is outbreeding depression (Lynch 1991, McClelland and Naish 2007), which is a reduction in fitness of the offspring of highly divergent genotypes (Lynch 1991), caused by disruption of co-adapted gene complexes (Orr 1996) and loss of local adaptation (Templeton et al. 1986).

In addition to inbreeding avoidance and longer-term evolutionary potential, the degree of adaptation of the initial founders to the new environment is a critical concern in

reintroductions. If early generations are poorly adapted to the new, local environment the populations may never become successfully established. One strategy to address this issue that also addresses risks of inbreeding and longer-term adaptation to changing conditions, is the adaptive potential strategy (Houde et al. 2015). This strategy aims to maximize genetic diversity among founders, such that the new population can rapidly adapt to conditions in the site of reintroduction. Alternatively, founders can be selected that are pre-adapted to, and have high fitness in, the reintroduction site; two strategies to achieve this outcome are ancestry matching and environment matching (Houde et al. 2015). In ancestry matching, a source population is selected that is most genetically similar to the extirpated population at the designated reintroduction location. This strategy requires information on the genetic make-up of the historic, extirpated populations. In environment matching, a source population is selected that occupies an environment most similar to that of the reintroduction location. This strategy requires detailed information on the habitat and environmental characteristics of both the reintroduction location and locations of potential source populations, as well as strong understanding of the key factors affecting survival and reproduction of individuals.

Butterflies are one of the insect taxa in which species declines have been widely documented (Sánchez-Bayo and Wyckhuys 2019), and many attempts have been made to reintroduce various butterfly species (Nakahama et al. 2022; Davis et al. 2021; Dincă et al 2018; Andersen et al. 2014; Soorae 2010; Wynhoff 1998). Very few earlier butterfly reintroductions were successful, which has been attributed to a lack of accepted guidelines or failure to follow recommended protocols, such as the recommendation to release >60 individuals sourced form a large heterogenous population to maintain genetic

diversity and thorough assessment of reintroduction habitat (Joyce and Pullin 2003). Failure of the reintroduction of the Miami Blue Butterfly (*Cyclargus thomasi bethunebakeri*), for example, is attributed to limited reintroduction sites, difficulty rearing larvae, and tropical cyclones at the reintroduction sites (Soorae 2010). In contrast, successful reintroduction of the Large Blue Butterfly (*Maculinea arion*) in England has been largely attributed to strict adherence to the reintroduction guidelines of the International Union for the Conservation of Nature (IUCN/SSC 2013). Indeed, 19 generations after reintroduction, Andersen et al. found the introduced population showed no loss of genetic diversity even though it represented a unique subset of genetic diversity compared to its source population (2014).

The Mottled Duskywing, *Erynnis martialis* (Scudder, 1870) is a medium-sized, brown butterfly belonging to the skipper family (Lepidoptera: *Hesperiidae*; *Pyrginae*; COSEWIC 2015). It is endemic to North America where it has become uncommon to rare. It was listed as endangered in Ontario, Canada in 2012, with threats to the populations including loss of habitat, pesticide use, host plant competition with invasive species, and climate change (COSEWIC 2015). Severe population declines have also been observed in the United States, where the species has become extirpated in some regions (COSEWIC 2012). As a part of the recovery plan for the Mottled Duskywing in Canada, the species is being reintroduced to Pinery Provincial Park in Grand Bend, Ontario, where the species historically occurred but has since been extirpated. Future reintroduction to Norfolk County is also being considered.

Though historical Mottled Duskywing specimens from the reintroduction site were available, I was not able to successfully extract DNA extraction from them. Therefore, the genetics of that historical population are unknown, and the adaptive potential strategy could not be considered. Additionally, no research to date aside from field observations has been published regarding environmental preferences of *Erynnis martialis*, except for presence of its host plants. Therefore, I focused on examining the genetic diversity and structure, as well as demographic stability, of the Mottled Duskywing populations in Ontario, Canada, to inform selection of a source population using the adaptative potential strategy. To analyze genetic structure and variation in the species more broadly, my analyses also included some of the nearest populations in the United States (NY, MI) as well as the other remaining Canadian populations in the province of Manitoba. Data from 21 polymorphic microsatellite markers were used to assess genetic diversity of potential source populations to evaluate different reintroduction scenarios. Additionally, I assessed genetic divergence and genetic structure among populations, as well as demographic stability of the populations through estimation of effective population sizes and bottlenecks, to inform Mottled Duskywing biology and conservation, more generally.

3.3 Methods

3.3.1 Sample collection

Twelve whole female adult Mottled Duskywing butterflies, originally from Marmora, Ontario were obtained from a captive rearing experiment at the Cambridge Butterfly Conservatory, after they had died in captivity. Upon their deaths and following arrival in the laboratory, they were stored in a -80 °C freezer.

In 2019, I collected non-lethal wing or leg samples from most known extant populations in Ontario, Canada including Burlington, Marmora, Oakville, and the Rice Lake Area (Figure 3-1). A population in Northern Michigan was also sampled in 2019. In 2020, I resampled some of these sites to assess temporal variation in allele frequencies and estimate genetic drift and obtained samples from additional sites in the Rice Lake and Burlington areas in Ontario, as well as in eastern New York State, and sites from southern Manitoba. I defined different populations as sites inhabited by *E. martialis* that are 2 km or farther apart; without prior knowledge on genetic variation in the species it was possible that some of the samples did not represent truly isolated populations, but this would be revealed by my analyses (Figure 3-1).

In total, 312 Mottled Duskywings were sampled (Table 3-1). At each site, butterflies were captured by netting and placed in a small jar. The jar was placed on ice in a cooler for approximately 15 min to calm the butterfly. After 15 min of cooling, the butterfly was removed from the jar and either a small (approximately 0.25 cm²) sample of wing tissue from each hind wing or a single hind leg was collected from each butterfly using forceps. After each individual was sampled, forceps were cleaned by wiping with

70 % isopropyl alcohol swabs before re-use. Wing and leg samples were immediately placed individually in Eppendorf tubes filled with absolute ethanol or filled with silica powder covered with a thin layer of cotton batting to avoid silica sticking to butterfly tissue. Lastly, prior to release, butterflies were marked with a small amount of acrylic paint on either the wing or the thorax to avoid re-sampling. Though no published information exists regarding longevity of paint marks in the field, individuals have been observed with paint marks up to 21 days after marking (Demarse, unpublished data). Upon arrival at the laboratory, all samples were stored at -80 °C. A maximum of 30 butterflies were sampled from each site per year to minimize negative impacts to at-risk populations. Removal of these 'wing clips' mimics regular wear and tear that would occur on the wings of a butterfly. Extensive research has shown that this method does not affect a butterfly's ability to fly, survive, or mate (Crawford et al. 2013; Hamm et al. 2010; Figure 3-2). Similarly, removal of a leg has been demonstrated to have no measurable effect on survival or behaviour of butterflies, even small, delicate species (Marschalek et al. 2013).



Figure 3-1. Locations of Mottled Duskywing populations sampled for population genetics analyses. Only general areas disclosed due to sensitivity of this information, considering the endangered status of the species. Created using ArcGIS Pro (ArcGIS 2010).

Table 3-1. Breakdown of Mottled Duskywing samples collected from all known populations with permission to sample from Ontario, southern Manitoba, eastern New York, and northeastern Michigan. Whole butterfly samples provided by the captive rearing program and non-lethal samples refer to wing clips or a leg collected in the field. Latitude and longitude, and more specific locations are not reported due to the sensitivity of this information given the endangered status of the species.

Site	Province or State	2019 whole butterflies	2019 non-lethal sample	2020 non-lethal sample	Total
Burlington	Ontario	0	26	30	56
Burlington 2	Ontario	0	5	0	5
Burlington 3	Ontario	0	0	2	2
Burlington Gas Line	Ontario	0	5	30	35
Burlington Hydro Corridor	Ontario	0	5	0	5
Manitoba	Manitoba	0	0	17	17
Marmora	Ontario	12	18	30	60
Michigan	Michigan	0	11	12	23
New York	New York	0	0	30	30
Oakville	Ontario	0	1	15	16
Rice Lake 1	Ontario	0	13	30	43
Rice Lake 2	Ontario	0	0	10	10
Rice Lake 3	Ontario	0	0	10	10
Total		12	84	216	312



Figure 3-2. An unsampled Mottled Duskywing in the wild that has experienced natural wing damage. The natural damage is similar in size and shape to the damage resulting from collection of a non-lethal wing clip removed for DNA extraction. Photo credit: John Christensen.

3.3.2 DNA Extraction

Genomic DNA was extracted with a DNeasy Blood and Tissue Kit (Qiagen, Germantown, MD), using a sterile pestle to crush and grind butterfly tissue for better lysis of cells. Extraction methods from whole butterflies are outlined in Chapter 2. For wing clips and legs, the entire sample was used. A single elution with 200 μ l AE buffer warmed to 36 °C was used to collect the DNA.

3.3.3 Microsatellite genotyping

Primers and multi-plex PCR protocols described in Chapter 2 were used to amplify microsatellites from all samples. All individuals were genotyped at a total of 24 loci that amplified cleanly and consistently and were variable (see Chapter 2). Microsatellite loci were amplified using one fluorescently labelled primer per pair (6-FAM, VIC, NED, or PET label), and sized by fragment analysis on a 3730S capillary DNA analyzer, using LIZ-500 size standard (Applied Biosystems, Foster City, CA), at the University of British Columbia's Sequencing and Bioinformatics Consortium (British Columbia, Canada). Genotypes were called using GeneMarker® HID software (SoftGenetics, State College, Pennsylvania, USA).

3.3.4 Data analysis

Samples taken from the same location in different years were analyzed separately as unique populations. All data analyses only included samples where more than 5 individual butterflies were sampled in a given year. All loci were checked for deviations from Hardy Weinberg Equilibrium (HWE) or for evidence of linkage disequilibrium, within each population, using GENEPOP'007 software (Rousset 2008). Additionally, loci were checked for evidence of null alleles using MICROCHECKER v2.2.3 (Van Oosterhout et al., 2004).

3.3.4.1 *Genetic diversity*

As metrics of genetic diversity, I calculated expected heterozygosity (H_E; the proportion of individuals in a population that are expected to be heterozygotes under HWE) and observed heterozygosity (H_o; the proportion of heterozygous loci across all loci in all individuals), for each sample, using R statistical software (R Core Team 2020) with packages 'adegenet' (Jombart 2008), 'pegas' (Paradis 2010), and 'hierfstat' (Goudet 2005). Expected heterozygosity is calculated using the equation $1 - \sum p_i^2$, where p_i is the frequency of the *i*th allele, and the operator is summation over all alleles (Jombart 2014). Additionally, I calculated allelic richness (A_R; number of alleles independent of sample size), using R statistical software (R Core Team 2020) and the package 'diveRsity' (Keenan et al. 2013). Allelic richness is calculated in this package using 1000 re-samples where n is the smallest sample in the input data file, with replacement per locus per population sample (Keenan 2017).

3.3.4.2 *Population structure*

I calculated Fixation Index (F_{ST} ; measure of population differentiation due to genetic structure; Weir and Cockerham 1984) using GENEPOP'007 software (Rousset 2008) for all population pairs. For sites where there were sufficient sample sizes for both 2019 and 2020 sampling years, only the sampling year with more individuals sampled was included in these analyses. F_{ST} ranges from 0 to 1, where 0 indicates no genetic differences and 1 is the maximum possible amount of genetic differentiation (Hartl and

Clark 1997). However, F_{ST} values also depend on the variability of the markers used and cannot often be compared between studies of different species (Meirmans and Hedrick 2011).

To test for isolation by distance (IBD), I assessed the correlation between pairwise genetic differentiation (F_{ST}) and log base 10 geographic distance (km) using a Mantel test implemented with the 'ecodist' package (Goslee and Urban 2007) in R statistical software (R Core Team 2020). I calculated straight-line geographic distances using Google Earth Pro software and log (base 10) transformed the distances due to the wide range (2.2 km–1867.4 km). Where samples were taken in two different years at a given site, I only included the larger of the two samples in the IBD analysis.

To identify and describe genetic clusters within the Mottled Duskywing, I ran a Discriminant Analysis of Principal Components (DAPC) using the 'adegenet' package (Jombart 2008) in R (R Core Team 2020). I used the *find.clusters* function to detect the number of clusters using K-means associated with the lowest Bayesian Information Criterion (BIC). I used the cross-validation function *xvalDapc* to confirm the number of retained principal components (PCs). Discriminant analyses of principal components is a multivariate method used to identify and describe groups of genetically similar individuals (Jombart et al. 2010). This method provides some benefits not provided by Bayesian clustering methods such as STRUCTURE, such as not relying on assumptions about type of population subdivision. In contrast to other multivariate analyses such as a principal component analysis (PCA), DAPC uses K-means clustering of principal components that focuses on between-group variability, while minimizing within-group variation (Liu and Zhao 2006). Therefore, a DAPC is a more effective at discrimination of individuals into groups (Jombart et al. 2010). Bayesian Information Criterion (BIC) are used to assess the best supported model or number of clusters. On simulated data, DAPC was found to be as accurate as STRUCTURE in detecting clusters, and better at capturing the structure in more complex population genetic scenarios, such as where there is more subtle hierarchical structure or isolation by distance (Jombart et al. 2010).

Additionally, to infer population structure further I used STRUCTURE 2.3 using the no admixture model and correlated allele frequencies (Pritchard et al. 2000; Falush et al. 2003). Like DAPC, STRUCTURE (Pritchard et al. 2000) aims to assign individuals to genetic clusters and has become one of the most widely used programs to assess genetic stratification (Earl and vonHoldt 2012). It is a model-based Bayesian clustering method where there are "K" populations, and each population is characterized by a set of allele frequencies at each locus (Pritchard et al 2000). Sampled individuals are probabilistically assigned to one or more population(s) depending on if they are admixed, in such a way as to minimize Hardy-Weinberg and linkage disequilibrium within the K populations. There are multiple different methods used to identify the optimal K (number of clusters), the most used being the ΔK method (Evanno et al. 2005). The ΔK method is based on the rate of change in the log probability of data between successive K values. While it is effective at identifying the uppermost level of hierarchical population structure, it can fail to detect finer structure and may not perform well when the sample sizes are uneven (Puechmaille 2016). This led to the alternative estimators: MedMedK, MedMeaK, MaxMedK, and MaxMeaK (Puechmaille 2016). These estimators are based on the count of the number of clusters that are contained in at least one subpopulation and were found
to be more accurate on both even and uneven datasets. STRUCTURE is most accurate when run at multiple replications, and because stochastic simulation is part of the inference, independent analyses of the same data can result in several distinct outcomes (Jakobsson and Rosenberg 2007). The computer program CLUMPP takes membership coefficient matrices from multiple STRUCTURE runs and outputs them permuted so that all replicates have as close a match as possible.

No specific mutation model is assumed in STRUCTURE, and it is appropriate for unlinked microsatellite data (Pritchard et al. 2009). I ran 10 replicates at each estimated population number or K (setting K to vary from 1 to 16), with a burn-in of 100,000 iterations and collection of data of 1,000,000 iterations. The upper limit of estimated population number was set to 16 as there were 14 sampled populations (with sites separated by sampling years) and an upper limit of slightly more than 14 would allow for detection of potential subpopulations within sites. The results from STRUCTURE were then processed in STRUCTURE HARVESTER (Earl and vonHoldt 2012) and STRUCTURE SELECTOR (Li and Liu 2018) and the highest means from MedMeaK, MaxMeaK, MedMedK, and MaxMedK were used to infer the optimal K (Puechmaille 2016). Finally, the output with the optimal selected number of populations (in this case, K = 9) was processed using CLUMPAK for averaging of assignment scores and visualization of STRUCTURE bar plots (Kopelman et al. 2015).

3.3.4.3 Demographic stability

I determined estimates of variance effective population sizes (N_e) using the linkage disequilibrium method (Waples and Do 2008), as implemented in NeEstimator V2.1 (Do et al. 2014), for populations with more than 5 individuals sampled for both 2019 and 2020 sampling years, as required by the software using base settings. NeEstimator assumes discreet, nonoverlapping generations (Do et al. 2014). Estimation of contemporary N_e using the linkage disequilibrium (LD) method has become an important tool for conservation genetics and is the most used method (Hollenbeck et al. 2016). This method estimates N_e based on temporal allele frequency difference; as a population decreases in effective size, allele frequencies will change more between generations due to random genetic drift and increased inbreeding (Caballero 1994). The difference in allele frequencies is used to estimate the idealized population size that could produce a change as large as the one observed.

I used the program BOTTLENECK (Cornuet and Luikart 1997) to detect evidence of recent bottlenecks in Mottled Duskywing populations, for sites with sample sizes larger than 15 individuals, as required by the program. This analysis is based on the expectation that by preferentially eliminating rare alleles, a bottleneck reduces allelic diversity more rapidly than it reduces heterozygosity (Nei 1978). As such, the analysis estimates the heterozygosity expected from a sample, given the observed allelic diversity, under a model of mutation-drift equilibrium, and compares that value statistically to the heterozygosity observed in the sample. An excess of observed heterozygosity, relative to the expected, may indicate a recent bottleneck. I used the two-phase mutational model, as recommended as most appropriate for microsatellites (Di Rienzo et al. 1994) using the sign probability test.

3.4 Results

Of the 24 microsatellite loci analyzed, none of the pairs of loci showed significant linkage disequilibrium in any population. Three loci (EMusat10, EMusat13, EMusat35) showed significant deviation from HWE in four or more populations (P<0.01) and showed evidence of null alleles in six populations (Table 3-2). These loci were removed from all further analyses and all subsequent results were based on the remaining 21 loci.

3.4.1 Genetic diversity

The expected heterozygosity of Mottled Duskywing populations ranged from 0.45 (Michigan and Oakville) to 0.59 (New York) (Table 3-3). The observed heterozygosity of Mottled Duskywing populations ranged from 0.39 (Oakville) to 0.53 (New York) (Table 3-3). Allelic richness of Mottled Duskywing populations ranged from 2.70 (Oakville) to 3.99 (Manitoba) (Table 3-3). All metrics of genetic diversity (H_E , H_O , A_R), remained relatively stable across generations for sites that were sampled both in 2019 and 2020.

Table 3-2. Mottled Duskywing microsatellite loci showing evidence of deviation from Hardy Weinberg Equilibrium and null alleles according to GENEPOP'007 software (Rousset 2008) and MICROCHECKER v2.2.3 (Van Oosterhout et al., 2004) respectively. Out of HWE refers to loci out of Hardy Weinberg Equilibrium and Evidence of Null refers to loci showing evidence of null alleles.

Locus	Out of HWE	Evidence of Null		
EMusat10	Burlington 2020, New York 2020	Marmora 2019 & 2020, Burlington 2019 & 2020,		
		Burlington Gas Line 2020, New York 2020		
EMusat13	Burlington 2019 & 2020, Rice Lake 1 2019 &	Marmora 2019, Burlington 2019 & 2020, Rice		
	2020, New York 2020, Manitoba 2020	Lake 1 2020, New York 2020, Manitoba 2020		
EMusat35	Marmora 2019 & 2020, Burlington 2019 &	Marmora 2019, Burlington 2019 & 2020,		
	2020, Burlington Gas Line 2020, Rice Lake 1	Burlington Gas Line 2020, Rice Lake 1 2020,		
	2020, Oakville 2020, Rice Lake 3 2020, Maritaba 2020	New York 2020		
	Wannoda 2020			

Table 3-3. Diversity metrics calculated using R statistical software (R Core Team 2020) with packages "adegenet" (Jombart 2008), "pegas" (Paradis 2010), "hierfstat" (Goudet 2005), and "diveRsity" (Keenan et al. 2013), from microsatellite data obtained from Mottled Duskywing populations in Ontario, Manitoba, Michigan, and New York.

Site	H _E	H _O	A _R
Marmora 2019	0.54	0.51	3.18
Marmora 2020	0.52	0.50	3.26
Rice Lake 1 2019	0.50	0.50	3.59
Rice Lake 1 2020	0.54	0.49	3.52
Rice Lake 2 2020	0.47	0.46	3.20
Rice Lake 3 2020	0.53	0.50	3.07
Burlington 2019	0.56	0.48	3.26
Burlington 2020	0.56	0.49	3.62
Burlington Gas Line 2020	0.55	0.48	3.63
Oakville 2020	0.45	0.39	2.70
Manitoba 2020	0.58	0.51	3.99
Michigan 2019	0.45	0.43	3.05
Michigan 2020	0.51	0.51	3.37
New York 2020	0.59	0.53	3.90

3.4.2 Population structure

3.4.2.1 Population differentiation

Fixation Index (F_{ST}) for population pairs ranged from 0.0182 (Rice Lake 1 and Rice Lake 2) to 0.2381 (Marmora and Rice Lake 3) (Table 3-4). Populations showed a significant correlation between log geographic distance and F_{ST} ($r_{45,43}$ =0.26, p<0.05; Figure 3-3). Low pairwise F_{ST} was observed between populations separated by less than approximately 8 km (i.e., log distance = 0.9 km), and pairwise F_{ST} values between populations separated by greater distances were much higher on average but showed no clear trend of increasing with increasing geographic distance.

calculated using genotypes at 21 microsatellite loci with GENEPOP'007 software (Rousset 2008).

 Site
 Burlington
 Burlington Gas Line
 Manitoba
 Marmora
 Michigan
 New York
 Oakville
 Rice Lake 1
 Rice Lake 2

 Burlington
 Burlington Gas Line
 Manitoba
 Marmora
 Michigan
 New York
 Oakville
 Rice Lake 1
 Rice Lake 2

Table 3-4. Pairwise FST between populations of Mottled Duskywing in Ontario, New York, Michigan, and Manitoba. FST values

		Ŭ			<u> </u>				
Burlington									
Burlington Gas Line	0.052								
Manitoba	0.119	0.081							
Marmora	0.147	0.152	0.173						
Michigan	0.116	0.134	0.144	0.155					
New York	0.110	0.115	0.106	0.155	0.101				
Oakville	0.170	0.167	0.191	0.222	0.207	0.150			
Rice Lake 1	0.132	0.126	0.126	0.191	0.166	0.080	0.168		
Rice Lake 2	0.163	0.151	0.170	0.193	0.188	0.099	0.161	0.018	
Rice Lake 3	0.176	0.174	0.198	0.238	0.237	0.134	0.142	0.069	0.044



Figure 3-3. The relationship between genetic differentiation (F_{ST}), estimated using 21 microsatellites, and log transformed geographic distance (km) between pairs of populations of the Mottled Duskywing in Ontario, New York, Michigan, and Manitoba.

3.4.2.2 Discriminant analyses of principal components

Using DAPC, the samples were sorted into nine suggested groups (Figure 3-4). The Marmora 2019 and 2020 samples clustered together strongly (Figure 3-4, group 1). Additionally, samples from both sampling years at Burlington, along with the Burlington Gas Line 2020 sample, were clustered together into two groups (3 and 7). The Rice Lake Area sampling sites (Rice Lake 1 2019, Rice Lake 1 2020, Rice Lake 2 2020, Rice Lake 3 2020) were clustered together into two groups (5 and 9). The USA sites (New York 2020, Michigan 2019, Michigan 2020) were also clustered together in two different groups (6 and 8). Oakville 2020 was differentiated from all other groups (4) as was Manitoba 2020 (2).



PC 1

Figure 3-4. Clustering of Mottled Duskywing samples, based on microsatellite data, using Discriminant Analysis of Principal Components (DAPC). Individuals are represented as dots and identified clusters represented by colour and inclusion of 95 % inertia ellipses placed on the plane using first two principal components.

3.4.2.3 STRUCTURE analysis

Like the DAPC analysis, STRUCTURE also grouped the samples into nine populations, based on MaxMeanK (Puechmaille 2016) (Figure 3-5). Marmora 2019 and 2020 were grouped together (Figure 3-5; blue). Rice Lake Plains samples were grouped together, although in two populations, one composed of N 2019 and 2020 (Figure 3-5, light pink) and the other composed of Rice Lake 2 2020 and Rice Lake 3 2020 (Figure 3-5, green), with some admixture between them. Both Burlington sampling years were grouped together (Figure 3-5, orange), with subtle mixing with the Burlington Gas Line 2020 sample (Figure 3-5, deep pink). All the USA samples were grouped together (Michigan 2019 and 2020, and New York 2020; Figure 3-5, purple). Oakville 2020 (Figure 3-5, light green) and Manitoba 2020 (Figure 3-5, bright pink) formed separate populations. The ninth cluster (Figure 3-5, yellow) is represented by subtle admixture across most populations.



Figure 3-5. Clustering of Mottled Duskywing samples from Ontario, New York, Michigan, and Manitoba, based on 21 microsatellite loci, using STRUCTURE (Pritchard et al. 2000; Falush et al. 2003), STRUCTURE SELECTOR (Li and Liu 2018), and CLUMPAK (Kopelman et al. 2015). Each vertical line represents a sampled individual and is divided into coloured segments representing admixture assignment to nine genetic clusters. RL denotes Rice Lake; B denotes Burlington; BGL denotes Burlington Gas Line.

3.4.3 Demographic stability

3.4.3.1 Effective population size

Estimated N_e ranged from 5.9–963.4 (Table 3-5). Estimates of "infinite" in the confidence intervals result when there is no evidence of drift, or temporal variation in allele frequencies once sampling error is accounted for (Waples and Do 2010). The largest effective population size estimated was for Burlington (approximately 963), followed by Marmora (approximately 35). Michigan and Rice Lake 1 had similar effective population sizes to each other, but considerably lower than the other two populations (approximately 6 for both).

Table 3-5. Variance in effective population sizes (N_e) for sampled Mottled Duskywing populations, with sample sizes above five individuals in both 2019 and 2020, based on microsatellite data, estimated by NeEstimator v2 (Do et al. 2014) using the linkage disequilibrium method (Waples and Do 2008). The last column presents the 95 % confidence intervals of effective population sizes.

Site	# of samples collected in 2019/2020	Estimated N _e (# of individuals)	95 % C.I. (# of individuals
Burlington	25/30	963.4	28.2-Infinite
Michigan	11/8	6.3	3.1-17.2
Rice Lake 1	13/30	5.9	3.4-10.7
Marmora	34/30	34.8	13.8-335.5

3.4.3.2 Bottleneck tests

Analyses using BOTTLENECK indicated no significant evidence of a recent bottleneck occurring in any of the populations tested (all p>0.05; Table 3-6).

Table 3-6. Tests for evidence of recent bottlenecks in Mottled Duskywing populations, for populations from which > 15 individuals were sampled. Analyses were conducted using BOTTLENECK software (Cornuet and Luikart 1997) under the two-phase model (Di Rienzo et al. 1994). 'Probability' indicates p-value determined by the sign test.

Site	Probability
Marmora 2019	0.34
Marmora 2020	0.21
Rice Lake 1 2020	0.45
Burlington 2019	0.13
Burlington 2020	0.09
Burlington Gas Line 2020	0.54
Oakville 2020	0.10
New York 2020	0.59
Manitoba 2020	0.57

3.5 Discussion

3.5.1 Genetic diversity

Overall, Mottled Duskywing populations had moderate levels of genetic diversity, compared to estimates obtained using microsatellite markers in other butterflies (Table 3-7). Among all Mottled Duskywing samples analyzed here, mean overall H_E was 0.525 and mean overall H_0 was 0.484. These values are similar to, or slightly lower than heterozygosity values observed in other butterfly species using microsatellite data (Table 3-3, Table 3-7). Mean overall A_R across all Mottled Duskywing samples was 3.381, which again is similar to, or slightly lower than estimates from other species (Table 3-3, Table 3-7). In general, species that are not threatened tend to have higher genetic diversity metrics compared to the threatened and endangered groups; Mottled Duskywing genetic diversity metrics are comparable with other threatened and endangered butterfly species (Table 3-3, Table 3-7). However, there are several threatened or endangered species that have fairly high genetic diversity estimates (e.g., Lycaena helle; Trense et al. 2021). This can occur if populations are large enough to buffer against genetic drift and inbreeding (Trense et al. 2021) or have a long history of population isolation such that they are better able to cope with small population size even if isolated (Habel and Schmitt 2012). Alternatively, high diversity rates might reflect historically high diversity that has not yet been lost due to a time-lag (Essl et al. 2015).

Table 3-7. Genetic diversity metrics reported from previous studies of butterfly species using microsatellites. H_E refers to expected heterozygosity, H_O refers to observed heterozygosity, and A_R refers to allelic richness. NR refers to metrics that were not reported in the article. I conducted a search on Google Scholar using the terms "allelic richness microsatellite butterfly" or "expected heterozygosity microsatellite butterfly" and selected 30 representative species that are categorized as either endangered, threatened or not threatened for comparison with genetic diversity levels in the Mottled Duskywing.

Species	Status	H _E	H _O	A _R	Study
Argynnis nerippe	Endangered	0.183-0.985	0.324-0.863	2.274-6.755	Jeong et al. 2018
Cyclargus thomasi bethunebakeri	Endangered	0.458-0.601	0.362-0.510	2.988-3.121	Saarinen et al. 2014
Euphydryas aurinia	Endangered	0.233-0.521	NR	2.212-1.148	Sigaard et al. 2008
Lycaena helle	Endangered	0.76-0.85	0.42 - 0.50	6.1–7.3	Trense et al. 2021
Melitaea protomedia	Endangered	0.396-0.662	0.434-0.600	2.22-3.47	Nakahama & Isagi 2017
Neonympha mitchellii francisci	Endangered	0.431-0.594	0.371-0.560	2.91-4.3	Milko et al. 2012
Oarisma poweshiek	Endangered	0.347-0.507	0.127 - 0.480	1.54-2.20	Saarinen et al. 2016
Parnassius apollo	Endangered	0.251-0.954	0.075-0.796	NR	Petenian et al 2005
Parnassius apollo filbricus	Endangered	0.46-0.60	0.53-0.59	3.05-4.05	Martínez et al. 2018
Phengaris arion	Endangered	0.611-0.742	0.500-0.705	4.33-7.51	Rutkowski et al. 2009
Zizina emelina	Endangered	0.06 - 1.00	0.06-0.85	NR	Sato et al. 2020
Boloria aquilonaris	Threatened	0.294-0.576	NR	0.888-1.445	Turlure et al. 2014
Danaus plexippus	Threatened	0.415-0.631	0.394-0.484	NR	Lyons et al. 2012
Erynnis propertius	Threatened	0.709-0.903	0.319-0.816	2.9-3.9	Zakharov & Hellmann 2007
Lycaena hippothoe	Threatened	0.64-0.78	0.44-0.66	5.0-7.2	Trense et al. 2021
Maculinea alcon	Threatened	0.32-0.65	NR	2.0-3.5	Vanden Broeck et al. 2017
Maculinea arion	Threatened	0.45-0.57	NR	NR	Andersen et al. 2014
Phengaris alcon	Threatened	0.46-0.61	0.42 - 0.67	3.10-5.89	Sielezniew et al. 2011
Phengaris nausithous	Threatened	0.70-0.71	NR	6.2–7.8	Ritter et al. 2013
Phengaris rebeli	Threatened	0.156-0.267	0.070-0.237	1.77-2.16	Rutkowski et al. 2009
Phengaris rebeli	Threatened	0.13-0.55	0.07 - 0.55	1.60-3.93	Sielezniew et al. 2011
Phengaris teleius	Threatened	0.41 - 0.67	NR	1.5-3.3	Ritter et al. 2013
Arhopala epimuta	Not threatened	0.690-0.778	0.362-0.441	7.16-8.51	Fauvelot et al. 2006
Erebia palarica	Not threatened	0.812-0.818	NR	6.7-8.39	Vila et al. 2009
Maniola jurtina	Not threatened	0.713-0.805	0.279-0.902	7.765–9.096	Greenwell et al. 2021
Papilio zelicaon	Not threatened	0.432-0.866	0.206-0.813	3.0-5.2	Zakharov & Hellmann 2007
Pararge aegeria	Not threatened	0.776-0.898	0.659-842	9.667-15.333	Vandewoestijne & Van Dyck 2010
Parnassius smintheus	Not threatened	0.72 - 0.78	NR	5.70-6.69	Keyghobadi et al. 2005
Polyommatus coridon	Not threatened	0.899–0.919	0.697 - 0.784	15.66-18.74	Habel et al. 2014
Polyommatus icarus	Not threatened	0.623-0.639	0.550-0.627	7.4-8.1	Piszter et al. 2021

There was not a large variation in the estimates of diversity among populations of Mottled Duskywing (Table 3-3), despite some sites being visibly much smaller than others and apparently containing many fewer butterflies (based on encounter and captures rates in the field). A potential explanation is that most smaller populations and patches are connected to larger ones through a metapopulation system, that allows them to maintain diversity via immigration despite being small (Aycrigg and Garton 2014). This may be the case for example in the Rice Lake area where only ten butterflies were captured at both Rice Lake 2 and Rice Lake 3, but diversity is still moderate and there is evidence of admixture with the larger population at Rice Lake 1 (Figure 3-4; Figure 3-5). Another potential explanation for moderate diversity in the smaller populations is that because habitat loss and decline have been recent, there is a time-lag in the loss of diversity, and populations have not yet reached an equilibrium level of diversity (Essl et al. 2015). Supporting this is the fact that $A_{\rm R}$ is more variable among the populations (lowest in the small, isolated population of Oakville); A_R approaches equilibrium faster than H_E and H_O , exemplified by A_R having stronger correlations with recent changes to population structure in other butterflies (Caplins et al. 2014).

3.5.2 Population structure

Though genetic differentiation and geographic distance were positively correlated, the correlation was not strong (Figure 3-3). This significant but weak IBD pattern appears to be due to low genetic differentiation of populations that are close together, within approximately 8 km specifically, and higher average differentiation of populations that are more than 8 km apart. Thus, rather than a gradual linear increase of differentiation with distance, there appears to be a threshold where populations more than 8 km apart are more highly differentiated from each other (Figure 3-3). It is also important to note that beyond the 8 km separation, populations no longer show any association between geographic and genetic distance and there is considerable variance in pairwise F_{ST} values. High mean and variability in pairwise genetic differentiation values, and a lack of association with geographic distance, are indicative of a dominant effect of genetic drift in determining genetic patterns and a lack of appreciable gene flow among populations (Hutchison & Templeton 1999). The IBD plot for the Mottled Duskywing therefore suggests a shift in the relative influence of genetic drift versus gene flow with spatial scale, such that gene flow between populations is influential at smaller distances below approximately 8-10 km, but genetic drift is dominant at larger scales (Hutchison & Templeton 1999). These IBD results were further supported in the DAPC and STRUCTURE analyses where it was primarily samples from the same locations representing different years, or sites near each other, that grouped together (Figure 3-4; Figure 3-5).

Genetic differentiation tends to reflect limited dispersal among populations (Bohonak 1999), so high differentiation, and limited or no gene flow, among populations further than 8 km apart suggest Mottled Duskywings have limited dispersal ability at that scale. Since most extant populations of Mottled Duskywing are separated by more than 8-10 km, they are likely isolated from each other. This limited dispersal ability, and wide separation of areas of suitable habitat, may limit the natural recolonization ability of *Erynnis martialis*, underscoring the importance of translocation and reintroduction for the species. On the other hand, given evidence of potential connectivity below approximately 8 km, future management plans for extant or reintroduced Mottled Duskywing populations may want to consider managing sites within an 8 km distance a single management unit. Management plans might also consider restoring habitat within 8 km of occupied sites to promote natural colonization and establishment of new populations. Furthermore, future studies should address how the abundance and distribution of habitat, and other landscape features, may affect movement and gene flow among populations at this scale.

An interesting result was the relatively low F_{ST} value (0.101) between the New York and Michigan populations, and their grouping in both DAPC and STRUCTURE analyses, despite an approximate 831 km distance (Figure 3-4; Figure 3-5). Contemporary dispersal between these two locations is very unlikely given their spatial separation. One possible explanation for their genetic similarity is that it is a legacy of historical connectivity, when populations and suitable habitats were much more widespread. However, high genetic differentiation among currently extant populations that are much closer together, such as Burlington and Marmora in Ontario, do not lend support to this hypothesis. The most parsimonious explanation for the apparent genetic similarity of these distant populations is simply that it is a consequence of the dominant effect of random genetic drift, which introduces high variation in pairwise population differentiation, independent of geographic distance (Hutchison and Templeton 1999). Homoplasy (identity in state but not by descent), which can occur due at microsatellites because of their high mutation rates and allele size constraints (Putman and Carborne 2014), may also be a contributing factor.

3.5.3 Demographic stability

Effective population sizes, for the four populations where I had sufficient sampling to estimate this parameter, were moderate to low, except for Burlington where the upper confidence interval indicated no evidence of drift after accounting for sampling error. A study of *Hipparchia semele*, a butterfly species of conservation concern that showed evidence of inbreeding, estimated an N_e range of 20–54 individuals using similar methods (De Ro et al. 2021). Effective population size estimates for the threatened *Maculinea alcon* butterfly using microsatellites were much lower at 1.6–17.6 individuals (Vanden Broeck et al. 2017). As the program required a substantial sample size for both sampling years, Mottled Duskywing N_e was only estimated for large sites where more individuals were sampled, so these results may be biased due to ease of sampling at sites with highest density of Mottled Duskywing. Overall, my results indicate that all but the largest populations of the Mottled Duskywing are likely subject to appreciable levels of genetic drift.

While I did not find significant evidence of recent bottlenecks in any populations, a review on genetic bottleneck testing suggests that these tests often fail to detect declines in populations, even when they are known to have occurred (Peery et al. 2012). The low power of bottleneck tests may be due to factors such as short duration of the bottleneck, immigration, and high pre-bottleneck genetic diversity, all of which can dampen or quickly eliminate the expected genetic signature (Peery et al. 2012). Although I cannot definitively rule out the occurrence of bottlenecks, particularly in populations for which sample sizes were too small for testing, my results do not point to recent bottlenecks as a basis for selection, or exclusion, of any potential source populations for reintroduction.

3.5.4 Recommendations for reintroduction

The reintroduction of the Mottled Duskywing to Pinery Provincial Park in Grand Bend, Ontario, has begun. The two main populations originally considered as a source were Marmora and Burlington. As the reintroduction was to occur in Ontario, an Ontario source population was preferred. Among known populations in the province, Marmora and Burlington seemed to have the highest abundance of Mottled Duskywing, based on the number of butterflies captured or observed during the flight season. Additionally, they were logistically good choices for reintroduction because they were easy to access, and permissions from landowners were readily obtained. From my study, and preliminary results that I had collected before the summer 2021 reintroduction to Pinery Provincial Park, I knew that both potential source populations held similar levels of genetic diversity. However, I also knew that they were highly differentiated from one another (based on pairwise F_{ST} and STRUCTURE analyses). Therefore, evaluating the genetic data I had obtained within the context of the adaptative potential strategy, I had recommended that either population would be a suitable source. For comparison, the previously mentioned successful reintroduction of *Maculinea arion* reported a mean estimated H_E of the reintroduced populations between 0.45–0.57, which was not significantly different from the H_E of the source population (Andersen et al. 2014). Estimated H_E for both Marmora (H_E = 0.52 in 2019 and H_E = 0.54 in 2020) and Burlington ($H_E = 0.56$ in both years) are within that range. Furthermore, given that each population had moderate levels of genetic diversity, but they were highly differentiated from each other, population mixing was not necessary or recommended. The recommendation to avoid mixing source populations was also supported by preliminary

testing I conducted showing evidence of infection by the reproductive parasite *Wolbachia* in both the Marmora and Burlington populations (Shayla Kroeze, unpublished data). In general, mixing of populations in reintroductions of the Mottled Duskywing should be avoided until further testing to determine the prevalence of *Wolbachia*, and the identity of strains in different populations, has been conducted (Dincă et al 2018).

It would be useful to continue attempts to extract DNA from historical specimens, where those are available, to determine the genetic make-up of extirpated populations that previously occupied proposed reintroduction sites. This information could inform founder selection based on ancestry match (Houde et al. 2015). In terms of environment match, an entire group of associated butterflies in the UK showed evidence of temperature-mediated local adaptation related to emergence timing (Roy et al. 2015). Mottled Duskywing have been observed to have one or two broods per year depending on geographic range (Layberry et al. 1998). Future reintroductions or population augmentation of the Mottled Duskywing in Ontario, such as the planned reintroduction to Norfolk County, should investigate environmental factors such as those that impact emergence timing that may influence genetic variation in Mottled Duskywings and may affect how well-adapted founder individuals are to local conditions in a reintroduction site. Important environmental characteristics may be subtle, as in the example of Hesperia comma, where the size of the host plant and the state of the surrounding ground cover were found to be important for the suitability of oviposition (Thomas et al. 1986). Detailed studies of habitat and environmental requirements may be particularly important for Mottled Duskywing considering populations have become extirpated from habitat that appears to be suitable, with abundant host plant and nectaring plants (personal obs.).

3.6 Conclusion

I used 24 microsatellite markers previously described in Chapter 2 to characterize genetic diversity, population structure, and demographic stability of Mottled Duskywing populations in Ontario and nearby locations. Overall, populations had moderate levels of genetic diversity. I also found evidence for high average genetic differentiation, and limited gene flow, among populations separated by more than approximately 8 km. I used this genetic information to inform the selection of the Marmora population as a suitable source population for reintroduction to Pinery Provincial Park.

3.7 References

- Andersen A, Simcox DJ, Thomas JA, Nash DR (2014). Assessing reintroduction schemes by comparing genetic diversity of reintroduced and source populations: A case study of the globally threatened large blue butterfly (*Maculinea arion*). *Biological Conservation* 175: 34–41. doi: 10.1016/j.biocon.2014.04.009.
- Apollonio M, Andersen R, Putman R (2010). European ungulates and their management in the 21st century. Cambridge: Cambridge University Press.
- ArcGIS [ArcGIS Pro]. Version 10.0. Redlands, CA: Environmental Systems Research Institute, Inc., 2010.
- Attard C, Möller L, Sasaki M, Hammer M, Bice C, Brauer C, Carvalho D, Harris J, Beheregaray L (2016). A novel holistic framework for genetic-based captivebreeding and reintroduction programs. *Conservation Biology* 30: 1060–1069. doi: 10.1111/cobi.12699.
- Aycrigg JL, Garton EO (2014). Linking metapopulation structure to elk population management in Idaho: a genetic approach. *Journal of Mammalogy* 95: 597–614. doi: 10.1644/12-MAMM-A-300.
- Bainbridge I (2014). PRACTITIONER'S PERSPECTIVE: How can ecologists make conservation policy more evidence based? Ideas and examples from a devolved perspective. *Journal of Applied Ecology* 51: 1153–1158. doi: 10.1111/1365-2664.12294.
- Bohonak AJ (1999). Dispersal, gene flow, and population structure. *The Quarterly Review of Biology* 74: 21–45. doi: 10.1086/392950.
- Brown RD (2013). The history of wildlife conservation in North America. In: Krausman P.R., Cain JWIII, editors. Wildlife Management and Conservation: Contemporary Principles and Practices. Baltimore: Johns Hopkins University Press. p. 6–23.
- Caballero A (1994). Developments in the prediction of effective population size. *Heredity* 73: 657–679. doi: 10.1038/hdy.1994.174.
- Caplins SA, Gilbert KJ, Ciotir C, Roland J, Matter SF, Keyghobadi N (2014). Landscape structure and the genetic effects of a population collapse. *Proceedings of the Royal Society Biological Sciences* 281: 20141798. doi: 10.1098/rspb.2014.1798.
- Cornuet JM, Luikart G (1997). Description and power analysis of two tests for detecting recent population bottlenecks from allele frequency data. *Genetics* 144: 2001–2014. doi: 10.1093/genetics/144.4.2001.
- COSEWIC (2012). COSEWIC Assessment and Status Report on the Mottled Duskywing *Erynnis martialis* in Canada. COSEWIC Committee on the Status of Endangered Wildlife in Canada. Retrieved January 19, 2022, from https://www.registrelepsararegistry.gc.ca/virtual_sara/files/cosewic/sr_hesperie_tachetee_mottled_dusky wing_1213_e.pdf.
- COSEWIC (2015). Recovery Strategy for the Mottled Duskywing (*Erynnis martialis*) in Ontario. Canada: Ontario Ministry of Natural Resources and Forestry. Ontario Recovery Strategy Series. Retrieved March 27, 2020, from https://collections.ola.org/mon/29008/331486.pdf.

- Crawford LA, Koscinski D, Watt KM, McNeil JN, Keyghobadi N (2013). Mating success and oviposition of a butterfly are not affected by non-lethal tissue sampling. *Journal of Insect Conservation* 17: 859–864. doi: 10.1007/s10841-013-9566-8.
- Daniels JC, Nordmeyer C, Runquist E (2018). Improving standards for at-risk butterfly translocations. *Diversity* 10: 67–80. doi: 10.3390/d10030067.
- Davis ML, Barker C, Powell I, Porter K, Ashton P (2021). Combining modelling, field data and genetic variation to understand the post-reintroduction population genetics of the Marsh Fritillary butterfly (*Euphydryas aurinia*). *Journal of Insect Conservation* 25: 875–886. doi: 10.1007/s10841-021-00354-3.
- De Ro A, Vanden Broeck A, Verschaeve L, Jacobs I, T'Jollyn F, Van Dyck H, Maes D (2021). Occasional long-distance dispersal may not prevent inbreeding in a threatened butterfly. *BMC Ecology and Evolution* 21: 224. doi: 10.1186/s12862-021-01953-z
- Deinet S, Ieronymidou C, McRae L, Burfield IJ, Foppen RP, Collen B, Böhm M (2013). Wildlife Comeback in Europe: the Recovery of Selected Mammal and Bird Species. Final report to rewilding Europe by ZSL. London: BirdLife International and the European Bird Census Council.
- Di Rienzo A, Peterson AC, Garza JC, Valdes AM, Slatkin M, Freimer NB (1994). Mutational processes of simple sequence repeat loci in human populations. *Proceedings of the National Academy of Sciences of the United States of America* 91: 3166–3170. doi: 10.1073/pnas.91.8.3166.
- Dincă V, Bálint Z, Vodă R, Dapporto L, Hebert PD, Vila R (2018). Use of genetic, climatic, and microbiological data to inform reintroduction of a regionally extinct butterfly. *Conservation Biology* 32: 828–837. doi: 10.1111/cobi.13111.
- Do C, Waples RS, Peel D, Macbeth GM, Tillett BJ, Ovenden JR (2014). NeEstimator V2: re-implementation of software for the estimation of contemporary effective population size (Ne) from genetic data. *Molecular Ecology Resources* 14: 209– 214. doi: 10.1111/1755-0998.12157.
- Drauch AM, Fisher BE, Latch EK, Fike JA, Rhodes OE Jr. (2008). Evaluation of a remnant lake sturgeon population's utility as a source for reintroductions in the Ohio River system. *Conservation Genetics* 9: 1195–1209. doi: 10.1007/s10592-007-9441-9.
- Earl DA, vonHoldt BM (2012). STRUCTURE HARVESTER: A website and program for visualizing STRUCTURE output and implementing the Evanno method. *Conservation Genetics Resources* 4: 359–361. doi: 10.1007/s12686-011-9548-7.
- Essl F, Dullinger S, Rabitsch W, Hulme PE, Pyšek P, Wilson JRU, Richardson DM (2015). Historical legacies accumulate to shape future biodiversity in an era of rapid global change. *Diversity and Distributions* 21: 534–547. doi: 10.1111/ddi.12312.
- Evanno G, Regnaut S, Goudet J (2005). Detecting the number of clusters of individuals using the software STRUCTURE: a simulation study. *Molecular Ecology* 14: 2611–2620. doi: 10.1111/j.1365-294X.2005.02553.x.
- Falush D, Stephens M, Pritchard JK (2003). Inference of population structure using multilocus genotype data: Linked loci and correlated allele frequencies. *Genetics* 164: 1567–1587. doi: 10.1111/j.1471-8286.2007.01758.x.

- Fauvelot C, Cleary DFR, Menken SBJ (2006). Short-Term Impact of 1997/1998 ENSOinduced disturbance on abundance and genetic variation in a tropical butterfly. *Journal of Heredity* 97: 367–380. doi: 10.1093/jhered/esl010
- Fraser DJ (2008). How well can captive breeding programs conserve biodiversity? A review of salmonids. *Evolutionary Applications* 1: 535–586. doi: 10.1111/j.1752-4571.2008.00036.x.
- Goslee SC, Urban DL (2007). The ecodist package for dissimilarity-based analysis of ecological data. *Journal of Statistical Software* 22: 1–19. doi: 10.18637/jss.v022.i07.
- Goudet J (2005). Hierfstat, a package for R to compute and test hierarchical F-statistics. *Molecular Ecology Notes* 5: 184–186. doi: 10.1111/j.1471-8286.2004.00828.x.
- Greenwell MP, Botham MS, Bruford MW, Day JC, Evans LC, Gibbs M, Middlebrook I, Roy DB, Watts K, Oliver TH (2021). The influence of chalk grasslands on butterfly phenology and ecology. *Ecology and Evolution* 11: 14521–14539. doi: 10.1002/ece3.8111.
- Groenen M, Archibald A, Uenishi H, et al. (2012). Analyses of pig genomes provide insight into porcine demography and evolution. *Nature* 491, 393–398. doi: 10.1038/nature11622.
- Habel, JC, Brückmann SV, Krauss J, Schwarzer J, Weig A, Husemann M, Steffan-Dewenter I (2015). Fragmentation genetics of the grassland butterfly *Polyommatus coridon*: Stable genetic diversity or extinction debt? *Conservation Genetics* 16: 549–558. doi: 10.1007/s10592-014-0679-8.
- Habel JC, Schmitt T (2012). The burden of genetic diversity. *Biological Conservation* 147: 270–274. doi: 10.1016/j.biocon.2011.11.028.
- Haig SM (1998). Molecular contributions to conservation. *Ecology* 79: 413–425. doi: 10.2307/176942.
- Hamm CA, Aggarwal D, Landis DA (2010). Evaluating the impact of non-lethal DNA sampling on two butterflies, *Vanessa cardui* and *Satyrodes eurydice*. *Journal of Insect Conservation* 14: 11–18. doi: 10.1007/s10841-009-9219-0.
- Hartl DL, Clark AG (1997). Principles of Population Genetics. Sinauer Associates, Inc. Sunderland, MA. 542 pp.
- Hedrick PW (2001). Conservation genetics: where are we now? *Trends in Ecology and Evolution* 16: 629–636. doi: 10.1016/S0169-5347(01)02282-0.
- Hollenbeck CM, Portnoy DS, Gold JR (2016). A method for detecting recent changes in contemporary effective population size from linkage disequilibrium at linked and unlinked loci. *Heredity* 117: 207–216. doi: 10.1038/hdy.2016.30.
- Houde ALS, Garner SR, Neff BR (2015). Restoring species through reintroductions: strategies for source population selection. *Restoration Ecology* 23: 746–753.
- Hughes KA, Sawby R (2004). Genetic variability and life-history evolution. Ferrie`re R, Dieckmann U, Couvet D (Eds.). Evolutionary conservation biology. Cambridge University Press, Cambridge.
- Hutchison DW, Templeton AR (1999). Correlation of pairwise genetic and geographic distance measures: inferring the relative influences of gene flow and drift on the distribution of genetic variability. *Evolution* 53: 1898–1914. doi: 10.1111/j.1558-5646.1999.tb04571.x.

- IUCN/SSC (2013). Guidelines for Reintroductions and Other Conservation Translocations. Version 1.0 IUCN Species Survival Commission, Gland, Switzerland viiii + 57 pp. Retrieved February 2, 2022, from https://portals.iucn.org/library/sites/library/files/documents/2017-065.pdf.
- Jakobsson M, Rosenberg NA (2007). CLUMPP: a cluster matching and permutation program for dealing with label switching and multimodality in analysis of population structure. *Bioinformatics* 23: 1801–1806. doi: 10.1093/bioinformatics/btm233.
- Jeong SY, Kim MJ, Kim SS, Kim I (2018). Development and validation of microsatellite markers for the endangered nerippe fritillary butterfly, *Argynnis nerippe* (Lepidoptera: Nymphalidae). *International Journal of Industrial Entomology* 37: 1–8. doi: 10.7852/ijie.2018.37.1.1.
- Jombart T (2008). adegenet: a R package for the multivariate analysis of genetic markers. *Bioinformatics*, 24: 1403–1405. doi: 10.1093/bioinformatics/btn129.
- Jombart T (2014). Introduction to population genetics analysis using R. Imperial College London MRC Centre for Outbreak Analysis and Modeling. Retrieved February 17, 2021, from https://adegenet.r-forge.r-project.org/files/montpellier/practical-MVAintro.1.0.pdf.
- Jombart T, Devillard S, Balloux F (2010). Discriminant analysis of principal components: a new method for the analysis of genetically structured populations. *BMC Genetics* 11: 94. doi: 10.1186/1471-2156-11-94.
- Joyce DA, Pullin AS (2003). Conservation implications of the distribution of genetic diversity at different scales: a case study using the marsh fritillary butterfly (*Euphydryas aurinia*). *Biological Conservation* 114: 453–461. doi: 10.1016/S0006-3207(03)00087-9.
- Keenan K (2017). Package 'diveRsity'. Retrieved February 18, 2022, from https://cran.rproject.org/web/packages/diveRsity/diveRsity.pdf.
- Keenan K, McGinnity P, Cross TF, Crozier WW, Prodöhl PA (2013). diveRsity: An R package for the estimation of population genetics parameters and their associated errors. *Methods in Ecology and Evolution* 4: 782–788. doi:10.1111/2041-210X.12067.
- Keyghobadi N, Roland J, Strobeck C (2005). Genetic differentiation and gene flow among populations of the alpine butterfly, *Parnassius smintheus*, vary with landscape connectivity. *Molecular Ecology* 14: 1897–1909. doi: 10.1111/j.1365-294X.2005.02563.x.
- Kopelman NM, Mayzel J, Jakobsson M, Rosenberg NA, Mayrose I (2015). CLUMPAK: A program for identifying clustering modes and packaging population structure inferences across K. *Molecular Ecology Resources* 15: 1179–1191. doi: 10.1111/1755-0998.12387.
- Layberry RA, Hall PW, Lafontaine JD (1998). The Butterflies of Canada. University of Toronto Press, Toronto. 280 pp.
- Lowe WH, Kovach RP, Allendorf FW (2017). Population genetics and demography unite ecology and evolution. *Trends in Ecology and Evolution* 32: 141–152. doi: 10.1016/j.tree.2016.12.002.

- Li YL, Liu JX (2018). StructureSelector: A web based software to select and visualize the optimal number of clusters using multiple methods. *Molecular Ecology Resources* 18: 176–177. doi: 10.1111/1755-0998.12719.
- Liu N, Zhao H (2006). A non-parametric approach to population structure inference using multilocus genotypes. *Human Genomics* 2: 353. doi: 10.1186/1479-7364-2-6-353.
- Lynch M (1991). The genetic interpretation of inbreeding depression and outbreeding depression. *Evolution* 45: 622–629. doi: 10.1111/j.1558-5646.1991.tb04333.x.
- Lyons JI, Pierce AA, Bibeau SM, Sternberg ED, Mongue AJ, De Roode JC (2012). Lack of genetic differentiation between monarch butterflies with divergent migration destinations. *Molecular Ecology* 21: 3433–3444. doi: 10.1111/j.1365-294X.2012.05613.x.
- Marschalek DA, Jesu JA, Berres ME (2013). Impact of non-lethal genetic sampling on the survival, longevity and behaviour of the Hermes copper (*Lycaena hermes*) butterfly. *Insect Conservation and Diversity* 6: 658–662. doi: 10.1111/icad.12024.
- Martínez JG, Mira Ó, Sánchez-Prieto CB, Barea-Azcón JM, Tinaut A (2018). Population size and genetic variability of a relict population of an endangered butterfly, *Parnassius apollo filabricus. Insect Conservation and Diversity* 11: 294–304. doi: 10.1111/icad.12276.
- McClelland E, Naish K (2007). What is the fitness outcome of crossing unrelated fish populations? A meta-analysis and an evaluation of future research directions. *Conservation Genetics* 8: 397–416. doi: 10.1007/s10592-006-9178-x.
- Meirmans PG, Hedrick PW (2011). Assessing population structure: FST and related measures. *Molecular Ecology Resources* 11: 5–18. doi: 10.1111/j.1755-0998.2010.02927.x.
- Milko LV, Haddad NM, Lance SL (2012). Dispersal via stream corridors structures populations of the endangered St. Francis' satyr butterfly (*Neonympha mitchellii francisci*). Journal of Insect Conservation 16: 263–273. doi: 10.1007/s10841-011-9413-8.
- Nakahama N, Hanaoka T, Itoh T, et al. (2022). Identification of source populations for reintroduction in extinct populations based on genome-wide SNPs and mtDNA sequence: a case study of the endangered subalpine grassland butterfly *Aporia hippia* (Lepidoptera; Pieridae) in Japan. *Journal of Insect Conservation* 26: 121– 130. doi: 10.1007/s10841-022-00369-4.
- Nakahama N, Isagi Y (2017). Recent transitions in genetic diversity and structure in the endangered semi-natural grassland butterfly, *Melitaea protomedia*, in Japan. *Insect Conservation and Diversity* 11: 330–340. doi: 10.1111/icad.12280.
- Nei M (1978). Molecular Evolutionary Genetics. Columbia University, New York.
- Oliver TH, Morecroft MD (2014). Interactions between climate change and land use change on biodiversity: attribution problems, risks, and opportunities. *WIREs Climate Change* 5: 317–335. doi: 10.1002/wcc.271.
- Orr HA (1996). Dobzhansky, Bateson, and the genetics of speciation. *Genetics* 144: 1331–1335. doi: 10.1093/genetics/144.4.1331.
- Paradis E (2010). pegas: an R package for population genetics with an integratedmodular approach. *Bioinformatics* 26: 419–420. doi: 10.1093/bioinformatics/btp696.

- Peery MZ, Kirby R, Reid BN, Stoelting R, Doucet-Bëer E, Robinson S, Vásquez-Carrillo C, Pauli JN, Palsbøll PJ (2012). Reliability of genetic bottleneck tests for detecting recent population declines. *Molecular Ecology* 21: 3403–3418. doi: 10.1111/j.1365-294X.2012.05635.x.
- Petenian F, Meglécz E, Genson G, Rasplus J-Y, Faure E (2005). Isolation and characterization of polymorphic microsatellites in *Parnassius apollo* and *Euphydryas aurinia* (Lepidoptera). *Molecular Ecology Notes* 5: 243–245. doi: 10.1111/j.1471-8286.2005.00891.x.
- Piszter G, Kertész K, Sramkó G, Krízsik V, Bálint Z, Biró LP (2021). Concordance of the spectral properties of dorsal wing scales with the phylogeographic structure of European male *Polyommatus icarus* butterflies. *Scientific Reports* 11: 16498. doi: 10.1038/s41598-021-95881-z.
- Puechmaille SJ (2016). The program structure does not reliably recover the correct population structure when sampling is uneven: subsampling and new estimators alleviate the problem. *Molecular Ecology Resources* 16: 608–627. doi: 10.1111/1755-0998.12512.
- Pritchard JK, Stephens M, Donnelly P (2000). Inference of population structure using multilocus genotype data. *Genetics* 155: 945–959. doi: 10.1093/genetics/155.2.945.
- Pritchard JK, Wen X, Falush D (2009). Documentation for structure software: Version 2.3. Retrieved February 2, 2022, from

https://www.ccg.unam.mx/~vinuesa/tlem09/docs/structure_doc.pdf.

- Putman AI, Carbone I (2014). Challenges in analysis and interpretation of microsatellite data for population genetic studies. *Ecology and Evolution* 4: 4399–4428. http://doi.org/10.1002/ece3.1305
- R Core Team (2020). Computing, R: A language and environment for statistical computing. R Foundation for Statistical.
- Ritter S, Michalski SG, Settele J, et al. (2013). Wolbachia infections mimic cryptic speciation in two parasitic butterfly species, *Phengaris teleius* and *P. nausithous* (Lepidoptera: Lycaenidae). *PLoS ONE* 8(11): e78107. doi: 10.1371/journal.pone.0078107.
- Rousset F (2008). GENEPOP ' 007: a complete re-implementation of the GENEPOP software for Windows and Linux. *Molecular Ecology Resources* 8: 103–106. doi: 10.1111/j.1471-8286.2007.01931.x.
- Roy DB, Oliver TH, Botham MS, et al. (2015). Similarities in butterfly emergence dates among populations suggest local adaptation to climate. *Global Change Biology* 21: 3313–3322. doi: 10.1111/gcb.12920
- Rutkowski R, Sielezniew M, Szostak A (2009). Contrasting levels of polymorphism in cross-amplified microsatellites in two endangered xerothermophilous, obligatorily myrmecophilous, butterflies of the genus Phengaris (*Maculinea*) (Lepidoptera: Lycaenidae). *European Journal of Entomology* 106: 457–469. doi: 10.14411/eje.2009.058.
- Sánchez-Bayo F, Wyckhuys KAG (2019). Worldwide decline of the entomofauna: A review of its drivers. *Biological Conservation* 232: 8–27. doi: 10.1016/j.biocon.2019.01.020.

- Sato D, Ueda S, Nakahama N, Izuno A, Isagi Y, Yago M, Hirai N (2020). Development of microsatellite markers for the endangered *Zizina Emelina* (de l'Orza, 1869) (Lepidoptera: Lycaenidae). *Japanese Journal of Environmental Entomology and Zoology* 31: 21–26. doi: 10.11257/jjeez.31.21.
- Seddon PJ, Armstrong DP, Maloney RF (2007). Developing the science of reintroduction biology. *Conservation Biology* 21: 303–312. doi: 10.1111/j.1523-1739.2006.00627.x.
- Sielezniew M, Rutkowski R, Ponikwicka-Tyszko D, Ratkiewicz M, Dziekańska I, Švitra G (2012). Differences in genetic variability between two ecotypes of the endangered myrmecophilous butterfly *Phengaris* (=*Maculinea*) *alcon* the setting of conservation priorities. *Insect Conservation and Diversity* 5: 223–236. doi: 10.1111/j.1752-4598.2011.00163.x.
- Sigaard P, Pertoldi C, Bo Madsen A, Søgaard B, Loeschcke V (2008). Patterns of genetic variation in isolated Danish populations of the endangered butterfly *Euphydryas aurinia*. *Biological Journal of the Linnean Society* 95: 677–687. doi: 10.1111/j.1095-8312.2008.01078.x.
- Soorae PS (2010). Global re-introduction perspectives: Additional case-studies from around the globe. IUCN/ SSC Re-introduction Specialist Group, Abu Dhabi, UAE.
- Templeton AR, Hemmer H, Mace G, Seal US, Shields WM, Woodruff DS (1986). Local adaptation, coadaptation, and population boundaries. *Zoo Biology* 5: 115– 125. doi: 10.1002/ZOO.1430050206.
- Thomas JA, Thomas CD, Simcox DJ, Clarke RT (1986). Ecology and declining status of the silver-spotted skipper butterfly (*Hesperia comma*) in Britain. *Journal of Applied Ecology* 23: 365–380. doi: 10.2307/2404023.
- Tracy LN, Wallis GP, Efford MG, Jamieson IG (2011). Preserving genetic diversity in threatened species reintroductions: how many individuals should be released? *Animal Conservation* 14: 439–446. doi: 10.1111/j.1469-1795.2011.00448.x.
- Trense D, Habel JC, Finger A, Fischer K (2021). Contrasting genetic responses to habitat fragmentation for two Lycaenid butterfly species. *Insect Conservation and Diversity* 1–11. doi: 10.1111/icad.12556.
- Turlure C, Vandewoestijne S, Baguette M (2014). Conservation genetics of a threatened butterfly: comparison of allozymes, RAPDs and microsatellites. *BMC Genetics* 15: 114. doi: 10.1186/s12863-014-0114-7.
- Van Oosterhout C, Hutchinson WFD, Wills DP, Shipley P (2004). MICRO-CHECKER: Software for identifying and correcting genotyping errors in microsatellite data. *Molecular Ecology Notes* 4: 535–538. doi: 10.1111/j.1471-8286.2004.00684.x.
- Vanden Broeck A, Maes D, Kelager A, Wynhoff I, WallisDeVries MF, Nash DR, Oostermeijer JGB, Van Dyck H, Mergeay J (2017). Gene flow and effective population sizes of the butterfly Maculinea alcon in a highly fragmented, anthropogenic landscape. *Biological Conservation* 209: 89–97. doi: 10.1016/j.biocon.2017.02.001.
- Vandewoestijne S, Van Dyck H (2010). Population genetic differences along a latitudinal cline between original and recently colonized habitat in a butterfly. *PLoS ONE* 5: e13810. doi: 10.1371/journal.pone.0013810.

- Vila M, Latasa T, Pino JJ, Verhulst G (2009). Characterization of ten polymorphic microsatellite markers for the endemic Chapman's ringlet, *Erebia palarica* (Lepidoptera: Nymphalidae). *European Journal of Entomology* 106: 485–490. doi: 10.14411/eje.2009.061.
- Waples RS, Do C (2008). LDNE: a program for estimating effective population size from data on linkage disequilibrium. *Molecular Ecology Resources* 8: 753–756. doi: 10.1111/j.1755-0998.2007.02061.x.
- Weir BS, Cockerham CC (1984) Estimating F-statistics for the analysis of population structure. *Evolution* 38: 1358–1370. doi: 10.1111/j.1558-5646.1984.tb05657.x.
- Werren J, Baldo L, Clark M (2008). *Wolbachia*: master manipulators of invertebrate biology. *Nature Reviews Microbiology* 6: 741–751. doi: 10.1038/nrmicro1969.
- Wynhoff I (1998). Lessons from the reintroduction of *Maculinea teleius* and *M. nausithous* in the Netherlands. *Journal of Insect Conservation* 2: 47–57. doi: 10.1023/A:1009692723056.
- Yang JB, Dong YR, Wong KM, Gu ZJ, Yang HQ, Li DZ (2018). Genetic structure and differentiation in *Dendrocalamus sinicus* (Poaceae: Bambusoideae) populations provide insight into evolutionary history and speciation of woody bamboos. *Scientific Reports* 8: 16933. doi: 10.1038/s41598-018-35269-8.
- Zakharov EV, Hellmann JJ (2008). Genetic differentiation across a latitudinal gradient in two co-occurring butterfly species: revealing population differences in a context of climate change. *Molecular Ecology* 17: 189–208. doi: 10.1111/j.1365-294X.2007.03488.x.

Chapter 4

4 General Discussion

I successfully developed microsatellite markers for the endangered Mottled Duskywing butterfly and characterized genetic diversity and structure of populations in Ontario, New York, Michigan, and Manitoba. My work represents the first research on Mottled Duskywing population genetics. Overall, populations had moderate levels of genetic diversity and, in most instances, were highly genetically differentiated from each other. This information was used to inform source population selection for the first reintroduction of this species, as part of its recovery strategy; in the summer flight season of 2021, over 700 adult butterflies, larvae, and pupae reared in captivity from females collected in the Marmora population were released to Pinery Provincial Park (Groleau 2021). The field crew at Pinery confirmed that reintroduced butterflies mated and laid eggs there (Ontario BSAR 2021). In addition to informing this reintroduction and future planned reintroductions, my research has filled key knowledge gaps about Mottled Duskywing biology and genetics. My results suggest that alternative source populations such as the Burlington or Rice Lake 1 populations could be used for future reintroductions to different locations in Ontario to prevent reducing genetic resources or causing excessive disturbance to the Marmora population, as they have comparable levels of genetic diversity.

4.1 Future applications and research

The genetic tools that I developed can now be used to monitor the genetic status of introduced and current populations. All reintroduced populations should be monitored according to IUCN guidelines, including genetic monitoring (IUCN/SSC 2013). Information from ongoing monitoring can inform the optimum number and size of further releases and identify unexpected threats to reintroduced populations. Estimation of key demographic and genetic parameters of a reintroduced population is critical to define reintroduction success and inform adaptive management strategies (DeMay et al. 2017).

The microsatellites I developed may also prove useful for selecting individual butterflies for captive rearing programs to ensure maximum genetic diversity. For example, adult females that provide eggs for the captive rearing program could be genotyped prior to any releases of their offspring. If any of these females are found to be very genetically similar, their offspring could be more effectively partitioned among different areas of a reintroduction site to distribute the genetic diversity and reduce inbreeding. Alternatively, a large number of potential egg-laying females could be collected in the wild and genotyped using non-lethal tissue samples before eggs are laid, and then only the most genetically diverse group retained for captive breeding. Given that captive rearing is very labour intensive, this approach could preserve time and resources used for captive breeding while maximizing genetic diversity of the reared offspring group.

In this thesis I made recommendations for source population selection based primarily on the adaptive potential strategy and ensuring sufficient genetic diversity among founders but balanced against considerations of outbreeding and potential effects of reproductive parasites. There is still scope to investigate alternate strategies for source population selection, specifically pre-existing adaptation strategies including environment matching and ancestry matching (Houde et al. 2015), for future reintroductions of the Mottled Duskywing. Future studies may investigate important environmental conditions for Mottled Duskywing survival, which would be necessary for developing any environment matching strategy. Building on such work, potentially functional or expressed genetic markers (e.g., SNPs) could be developed to allow research strategies such as genome-environment association (GEA) analyses that link functional genes to individual environmental predictors (Pluess et al. 2016). Considering Mottled Duskywing populations had moderate levels of genetic diversity, identification of large numbers of SNPs may be possible from even a small number of individuals sampled from a single population, using methods such as genotyping by sequencing (GBS; Deschamps et al. 2012) or restriction associated DNA sequencing (RADSeq: Andrews et al. 2016). These data could then be used to develop assays for moderate numbers (e.g., hundreds) of SNPs that could be genotyped using the small amounts of DNA obtained from non-lethal tissue samples. Though no GEA analyses have been conducted for butterfly species, there is evidence that fire regimes increased genetic diversity in two butterfly species (Gates et al. 2021). This study may be especially relevant to the Mottled Duskywing that is known to have a relationship with fire as the host plant will decline due to succession in the absence of disturbance (COSEWIC 2015; COSEWIC 2012).

Research that continues to attempt to extract DNA from, and genotype, historical specimens should also be further explored. Information about historical allele frequencies of Mottled Duskywings, especially in populations that have become extirpated from sites that are potential targets of reintroduction, would be necessary for ancestry matching. A combination of strategies could also be employed where if multiple potential source populations exhibit evidence for pre-existing adaption, either through environment or ancestry match, the population that also has a higher level of genetic diversity would be selected. Also, research to determine *Wolbachia* prevalence, and characterize strains of *Wolbachia*, in different populations is important for a better understanding of the risks associated with mixing individuals from different populations, either for sourcing reintroductions or augmenting declining populations.

4.2 Conservation implications

Considering that Mottled Duskywing populations show moderate levels of genetic diversity, conservation efforts directed towards extant populations might focus on threats other than low genetic diversity, such as habitat degradation, climate change, and pesticide use (COSEWIC 2015). Augmentation of current populations may not be necessary, although this view may be biased considering analysis was often not possible at sites where very few individuals were sampled, likely due to low butterfly abundance. The Oakville population exhibited the lowest amount of genetic diversity of the Ontario populations based on all three metrics (H_E , H_O , A_R). However, there are existing plans to construct a road within critical habitat for that population. If possible, nearby habitat should be restored, and the population could be augmented with butterflies from other

populations so the population can persist. However, considering the Oakville population has high genetic differentiation from all other sites and we still do not have information about *Wolbachia* variation in Mottled Duskywing populations, augmentation is only recommended if the population is on the brink of extinction.

Most populations I studied were genetically isolated from each other, meaning that each population represents a unique subset of the total genetic diversity of the Mottled Duskywing. This highlights the importance of conserving all existing populations to conserve overall genetic diversity. Additionally, persistence of different subsets of diversity across populations increase the chances of finding potential future matches (populations that have similar allele frequencies and/or exist in similar environments) if the pre-existing adaptation strategy were to be employed for future reintroductions (Houde et al. 2015).

Several studies have looked at population genetics of butterfly species postreintroduction (Davis et al. 2021; Kuussaari et al. 2015; Andersen et al. 2014; Schmitt et al. 2005; Irmgard 2001). Fewer studies have examined butterfly population genetics to inform source population selection prior to reintroduction, one making recommendations based on ancestry matching (Saarinen and Daniels 2012), and one based on selecting the closest population geographically, or environment matching (Gunson 2019), two different approaches within the pre-existing adaptation strategy. Future results from the reintroduction of the Mottled Duskywing will provide critical information on the success of the adaptive potential strategy for selection of a source population. I hope that my work can serve as a model for similar conservation and reintroduction projects on other
species, especially insect species that make up so much of the Earth's diversity (Stork 2018; Prather et al. 2013).

4.3 References

- Andersen A, Simcox DJ, Thomas JA, Nash DR (2014). Assessing reintroduction schemes by comparing genetic diversity of reintroduced and source populations: A case study of the globally threatened large blue butterfly (*Maculinea arion*). *Biological Conservation* 175: 34–41. doi: 10.1016/j.biocon.2014.04.009.
- Andrews KR, Good JM, Miller MR, Luikart G, Hohenlohe PA (2016). Harnessing the power of RADseq for ecological and evolutionary genomics. *Nature Reviews Genetics* 17: 81–92. doi: 10.1038/nrg.2015.28.
- COSEWIC (2015). Recovery Strategy for the Mottled Duskywing (*Erynnis martialis*) in Ontario. Canada: Ontario Ministry of Natural Resources and Forestry. Ontario Recovery Strategy Series. Retrieved March 27, 2020, from https://collections.ola.org/mon/29008/331486.pdf.
- COSEWIC (2012). COSEWIC Assessment and Status Report on the Mottled Duskywing *Erynnis martialis* in Canada. COSEWIC Committee on the Status of Endangered Wildlife in Canada. Retrieved January 19, 2022, from https://www.registrelepsararegistry.gc.ca/virtual_sara/files/cosewic/sr_hesperie_tachetee_mottled_dusky wing_1213_e.pdf.
- Davis ML, Barker C, Powell I, Porter K, Ashton P (2021). Combining modelling, field data and genetic variation to understand the post-reintroduction population genetics of the Marsh Fritillary butterfly (*Euphydryas aurinia*). *Journal of Insect Conservation* 25: 875–886. doi: 10.1007/s10841-021-00354-3.
- DeMay SM, Becker PA, Rachlow JL, Waits LP (2017). Genetic monitoring of an endangered species recovery: demographic and genetic trends for reintroduced pygmy rabbits (*Brachylagus idahoensis*). Journal of Mammalogy 98: 350–364. doi: 10.1093/jmammal/gyw197.
- Deschamps S, Llaca V, May GD. Genotyping-by-Sequencing in plants. *Biology* 1: 460–483. doi: 10.3390/biology1030460.
- Gates D, Jackson B, Schoville SD (2021). Impacts of fire on butterfly genetic diversity and connectivity. *Journal of Heredity* 112: 367–376. doi: 10.1093/jhered/esab027.
- Groleau C (2021). Endangered Mottled Duskywing butterfly making comeback thanks to dedicated scientist team. CBC News Kitchener-Waterloo, Ontario, Canada. Retrieved February 20, 2022, from https://www.cbc.ca/news/canada/kitchenerwaterloo/duskywing-butterfly-recovery-pinery-park-cambridge-butterflyconservatory-1.6137538.
- Gunson LR (2019). Genetic population structure of the Scotch argus butterfly (*Erebia aethiops*) in Britain: implications for conservation and future reintroductions (Masters dissertation). Lancaster University, Lancaster, England.
- Houde ALS, Garner SR, Neff BR (2015). Restoring species through reintroductions: strategies for source population selection. *Restoration Ecology* 23: 746–753. doi: 10.1111/rec.12280.
- Irmgard I (2001). At home on foreign meadows: the reintroduction of two *Maculinia* butterfly species (Doctoral dissertation). Wageningen University and Research, Wageningen, The Netherlands.

- IUCN/SSC (2013). Guidelines for Reintroductions and Other Conservation Translocations. Version 1.0 IUCN Species Survival Commission, Gland, Switzerland viiii + 57 pp. Retrieved February 2, 2022, from https://portals.iucn.org/library/sites/library/files/documents/2017-065.pdf.
- Kuussaari M, Heikkinen RK, Heliölä J, Luoto M, Mayer M, Rytteri S, von Bagh P (2015). Successful translocation of the threatened Clouded Apollo butterfly (*Parnassius mnemosyne*) and metapopulation establishment in southern Finland. *Biological Conservation* 190: 51–59. doi: 10.1016/j.biocon.2015.05.011.
- Ontario BSAR (2021). News and Events. Ontario Butterfly Species at Risk Recovery Team Ontario, Canada. Retrieved February 20, 2022, from https://www.onbutterflysar.com/news-and-events/.
- Pluess AR, Frank A, Heiri C, Lalagüe H, Vendramin GG, Oddou-Muratorio S (2016). Genome–environment association study suggests local adaptation to climate at the regional scale in *Fagus sylvatica*. *New Phytologist Foundation* 210: 589–601. doi: 10.1111/nph.13809.
- Prather CM, Pelini SL, Laws A, Rivest E, Woltz M, Bloch CP, Del Toro I, Ho C-K, Kominoski J, Newbold TAS, Parsons S, Joern A (2013). Invertebrates, ecosystem services and climate change. *Biological Reviews* 88: 327–348. doi: 10.1111/brv.12002.
- Saarinen EV, Daniels JC (2012). Using museum specimens to assess historical distribution and genetic diversity in an endangered butterfly. *Animal Biology* 62: 337–350. doi: 10.1163/157075612X624176.
- Schmitt T, Cizek O, Konvicka M (2005). Genetics of a butterfly relocation: large, small and introduced populations of the mountain endemic *Erebia epiphron silesiana*. *Biological Conservation* 123: 11–18. doi: 10.1016/j.biocon.2004.09.018.
- Stork NE (2018). How many species of insects and other terrestrial arthropods are there on Earth? Annual Review of Entomology 63: 31–45. doi: 10.1146/annurev-ento-020117-043348.

5 Appendices

Appendix A: Number of observed alleles per microsatellite locus per Mottled Duskywing population in Ontario, New York, Michigan, and Manitoba

Locus	Population	# of alleles
EMusat2	Burlington 2019	5
	Burlington 2020	5
	Burlington Gas Line 2020	6
	Manitoba 2020	6
	Marmora 2019	4
	Marmora 2020	3
	Michigan 2019	2
	Michigan 2020	5
	New York 2020	6
	Oakville 2020	5
	Rice Lake 1 2019	6
	Rice Lake 1 2020	5
	Rice Lake 2 2020	4
	Rice Lake 3 2020	5
EMusat3	Burlington 2019	2
	Burlington 2020	2
	Burlington Gas Line 2020	4
	Manitoba 2020	4
	Marmora 2019	5
	Marmora 2020	5
	Michigan 2019	5
	Michigan 2020	4
	New York 2020	4
	Oakville 2020	3
	Rice Lake 1 2019	3
	Rice Lake 1 2020	4
	Rice Lake 2 2020	3
	Rice Lake 3 2020	4

EMusat4	Burlington 2019	7
	Burlington 2020	7
	Burlington Gas Line 2020	8
	Manitoba 2020	11
	Marmora 2019	9
	Marmora 2020	8
	Michigan 2019	5
	Michigan 2020	7
	New York 2020	10
	Oakville 2020	3
	Rice Lake 1 2019	6
	Rice Lake 1 2020	7
	Rice Lake 2 2020	5
	Rice Lake 3 2020	0
EMusat5	Burlington 2019	2
	Burlington 2020	2
	Burlington Gas Line 2020	3
	Manitoba 2020	2
	Marmora 2019	2
	Marmora 2020	2
	Michigan 2019	1
	Michigan 2020	1
	New York 2020	3
	Oakville 2020	2
	Rice Lake 1 2019	1
	Rice Lake 1 2020	2
	Rice Lake 2 2020	1
	Rice Lake 3 2020	2
EMusat8	Burlington 2019	12
	Burlington 2020	13
	Burlington Gas Line 2020	14
	Manitoba 2020	11
	Marmora 2019	7
	Marmora 2020	7
	Michigan 2019	11
	Michigan 2020	11
	New York 2020	16
	Oakville 2020	6
	Rice Lake 1 2019	10
	Rice Lake 1 2020	12
	Rice Lake 2 2020	9
	Rice Lake 3 2020	6

EM	lusat9	Burlington 2019	3
		Burlington 2020	5
		Burlington Gas Line 2020	4
		Manitoba 2020	3
		Marmora 2019	3
		Marmora 2020	4
		Michigan 2019	2
		Michigan 2020	3
		New York 2020	3
		Oakville 2020	3
		Rice Lake 1 2019	4
		Rice Lake 1 2020	3
		Rice Lake 2 2020	3
		Rice Lake 3 2020	4
EM	lusat10	Burlington 2019	4
		Burlington 2020	5
		Burlington Gas Line 2020	5
		Manitoba 2020	6
		Marmora 2019	5
		Marmora 2020	6
		Michigan 2019	4
		Michigan 2020	5
		New York 2020	5
		Oakville 2020	4
		Rice Lake 1 2019	5
		Rice Lake 1 2020	5
		Rice Lake 2 2020	5
		Rice Lake 3 2020	5
EM	lusat11	Burlington 2019	3
		Burlington 2020	2
		Burlington Gas Line 2020	4
		Manitoba 2020	2
		Marmora 2019	2
		Marmora 2020	2
		Michigan 2019	1
		Michigan 2020	3
		New York 2020	2
		Oakville 2020	2
		Rice Lake 1 2019	3
		Rice Lake 1 2020	4
		Rice Lake 2 2020	2
		Rice Lake 3 2020	2

EMusat12	Burlington 2019	3
	Burlington 2020	2
	Burlington Gas Line 2020	3
	Manitoba 2020	4
	Marmora 2019	2
	Marmora 2020	2
	Michigan 2019	2
	Michigan 2020	2
	New York 2020	6
	Oakville 2020	1
	Rice Lake 1 2019	3
	Rice Lake 1 2020	2
	Rice Lake 2 2020	1
	Rice Lake 3 2020	1
EMusat13	Burlington 2019	2
	Burlington 2020	3
	Burlington Gas Line 2020	3
	Manitoba 2020	4
	Marmora 2019	3
	Marmora 2020	2
	Michigan 2019	2
	Michigan 2020	3
	New York 2020	2
	Oakville 2020	3
	Rice Lake 1 2019	3
	Rice Lake 1 2020	3
	Rice Lake 2 2020	2
	Rice Lake 3 2020	3
EMusat14	Burlington 2019	3
	Burlington 2020	2
	Burlington Gas Line 2020	4
	Manitoba 2020	3
	Marmora 2019	4
	Marmora 2020	4
	Michigan 2019	2
	Michigan 2020	3
	New York 2020	4
	Oakville 2020	4
	Rice Lake 1 2019	2
	Rice Lake 1 2020	3
	Rice Lake 2 2020	3
	Rice Lake 3 2020	3

EMusat16	Burlington 2019	3
	Burlington 2020	4
	Burlington Gas Line 2020	5
	Manitoba 2020	4
	Marmora 2019	3
	Marmora 2020	4
	Michigan 2019	3
	Michigan 2020	2
	New York 2020	5
	Oakville 2020	2
	Rice Lake 1 2019	4
	Rice Lake 1 2020	4
	Rice Lake 2 2020	4
	Rice Lake 3 2020	3
EMusat17	Burlington 2019	3
	Burlington 2020	3
	Burlington Gas Line 2020	3
	Manitoba 2020	3
	Marmora 2019	2
	Marmora 2020	3
	Michigan 2019	3
	Michigan 2020	2
	New York 2020	2
	Oakville 2020	4
	Rice Lake 1 2019	2
	Rice Lake 1 2020	2
	Rice Lake 2 2020	1
	Rice Lake 3 2020	2
EMusat18	Burlington 2019	7
	Burlington 2020	7
	Burlington Gas Line 2020	6
	Manitoba 2020	9
	Marmora 2019	6
	Marmora 2020	4
	Michigan 2019	3
	Michigan 2020	4
	New York 2020	7
	Oakville 2020	4
	Rice Lake 1 2019	6
	Rice Lake 1 2020	6
	Rice Lake 2 2020	6
	Rice Lake 3 2020	4

EMusat20	Burlington 2019	4
	Burlington 2020	4
	Burlington Gas Line 2020	3
	Manitoba 2020	4
	Marmora 2019	5
	Marmora 2020	2
	Michigan 2019	3
	Michigan 2020	3
	New York 2020	4
	Oakville 2020	2
	Rice Lake 1 2019	2
	Rice Lake 1 2020	2
	Rice Lake 2 2020	2
	Rice Lake 3 2020	2
EMusat22	Burlington 2019	3
	Burlington 2020	3
	Burlington Gas Line 2020	3
	Manitoba 2020	5
	Marmora 2019	4
	Marmora 2020	4
	Michigan 2019	3
	Michigan 2020	2
	New York 2020	3
	Oakville 2020	5
	Rice Lake 1 2019	4
	Rice Lake 1 2020	3
	Rice Lake 2 2020	2
	Rice Lake 3 2020	6
EMusat25	Burlington 2019	2
	Burlington 2020	4
	Burlington Gas Line 2020	2
	Manitoba 2020	3
	Marmora 2019	2
	Marmora 2020	2
	Michigan 2019	3
	Michigan 2020	2
	New York 2020	4
	Oakville 2020	5
	Rice Lake 1 2019	3
	Rice Lake 1 2020	3
	Rice Lake 2 2020	2
	Rice Lake 3 2020	3

		-
EMusat27	Burlington 2019	3
	Burlington 2020	3
	Burlington Gas Line 2020	3
	Manitoba 2020	3
	Marmora 2019	2
	Marmora 2020	2
	Michigan 2019	2
	Michigan 2020	2
	New York 2020	2
	Oakville 2020	1
	Rice Lake 1 2019	2
	Rice Lake 1 2020	2
	Rice Lake 2 2020	2
	Rice Lake 3 2020	2
EMusat31	Burlington 2019	6
	Burlington 2020	7
	Burlington Gas Line 2020	5
	Manitoba 2020	9
	Marmora 2019	6
	Marmora 2020	7
	Michigan 2019	5
	Michigan 2020	5
	New York 2020	9
	Oakville 2020	2
	Rice Lake 1 2019	6
	Rice Lake 1 2020	6
	Rice Lake 2 2020	5
	Rice Lake 3 2020	3
EMusat34	Burlington 2019	8
	Burlington 2020	10
	Burlington Gas Line 2020	8
	Manitoba 2020	10
	Marmora 2019	5
	Marmora 2020	6
	Michigan 2019	5
	Michigan 2020	5
	New York 2020	9
	Oakville 2020	5
	Rice Lake 1 2019	7
	Rice Lake 1 2020	7
	Rice Lake 2 2020	6
	Rice Lake 3 2020	6
	1000 Lake 5 2020	0

EMusat35	Burlington 2019	3
	Burlington 2020	5
	Burlington Gas Line 2020	4
	Manitoba 2020	4
	Marmora 2019	5
	Marmora 2020	6
	Michigan 2019	2
	Michigan 2020	3
	New York 2020	6
	Oakville 2020	5
	Rice Lake 1 2019	3
	Rice Lake 1 2020	4
	Rice Lake 2 2020	3
	Rice Lake 3 2020	3
EMusat36	Burlington 2019	9
	Burlington 2020	8
	Burlington Gas Line 2020	8
	Manitoba 2020	7
	Marmora 2019	5
	Marmora 2020	7
	Michigan 2019	6
	Michigan 2020	5
	New York 2020	8
	Oakville 2020	6
	Rice Lake 1 2019	8
	Rice Lake 1 2020	9
	Rice Lake 2 2020	8
	Rice Lake 3 2020	6
EMusat37	Burlington 2019	6
	Burlington 2020	6
	Burlington Gas Line 2020	5
	Manitoba 2020	8
	Marmora 2019	5
	Marmora 2020	6
	Michigan 2019	6
	Michigan 2020	6
	New York 2020	6
	Oakville 2020	4
	Rice Lake 1 2019	5
	Rice Lake 1 2020	6
	Rice Lake 2 2020	6
	Rice Lake 3 2020	5

EMusat40	Burlington 2019	7
	Burlington 2020	8
	Burlington Gas Line 2020	10
	Manitoba 2020	10
	Marmora 2019	6
	Marmora 2020	7
	Michigan 2019	8
	Michigan 2020	6
	New York 2020	10
	Oakville 2020	9
	Rice Lake 1 2019	7
	Rice Lake 1 2020	10
	Rice Lake 2 2020	7
	Rice Lake 3 2020	6

Curriculum Vitae

Name:	Shayla Kroeze
Post-secondary Education and Degrees:	The University of Western Ontario London, Ontario, Canada 2015-2019 B.Sc. Environmental Science and Biology
	The University of Western Ontario London, Ontario, Canada 2019-In Progress M.Sc. Biology
Honours and Awards:	Dean's Honour List 2017-2019
	Queen Elizabeth II Graduate Scholarship in Science and Technology (Declined) 2020
	Natural Sciences and Engineering Research Council of Canada (NSERC) Canada Graduate Scholarship 2020-2021
	Karen Auzins Scholarship The University of Western Ontario Biology Graduate Education Committee 2021
Related Work Experience	Laboratory and Field Assistant The University of Western Ontario 2017-2019
	Site Project Manager (Independent Contract) Natural Resource Solutions Inc. 2019-2020
	Teaching Assistant The University of Western Ontario 2019-2022

Publications:

Hickey, M.B.C, Filion, J., Brown, K.A.M., and **Kroeze, S.L.** (2017). Monitoring the efficacy of bat houses as roosting habitat enhancement for Little Brown Myotis (*Myotis lucifugus*). Technical report. Prepared for the Ontario Ministry of Natural Resources and Forestry, pp. 1-11.

Hickey, M.B.C, MacDonald-Ware, J., **Kroeze, S.L.**, and Brown, K.A.M.T. (2017). Efficacy of kiosks as replacement habitat for Barn Swallows. Technical report. Prepared for the Ontario Ministry of Transportation, pp. 1-34.

Hanna-Benson, C., **Kroeze, S.**, Gandhi, R., Haffie, T., and Wahl, L.M. (2020). Students as partners in collaborative course design and educational research. *International Journal for Students As Partners*, 4(2), 61–80. https://doi.org/10.15173/ijsap.v4i2.4237

Alshwairikh, Y.A., **Kroeze, S.L.**, Olsson, J., Stephens-Cardenas, S.A., Swain, W.L., Waits, L.P., Horn, R.L., Narum, S.R., and Seaborn, T. (2021). Influence of environmental conditions at spawning sites and migration routes on adaptive variation and population connectivity in Chinook salmon. *Ecology and Evolution*, 11, 16890–16908. https://doi.org/10.1002/ece3.8324